

Engineering Reports

RESEARCH ARTICLE OPEN ACCESS

MaOAOA: A Novel Many-Objective Arithmetic Optimization Algorithm for Solving Engineering Problems

Pradeep Jangir^{1,2,3,4} | Arpita⁵ | Sundaram B. Pandya⁶ | Gulothungan G.⁷ | Mohammad Khishe^{8,9} ^[b] | Bhargavi Indrajit Trivedi¹⁰

¹University Centre for Research and Development, Chandigarh University, Gharuan, Mohali, India | ²Department of CSE, Graphic Era Hill University, Dehradun, India | ³Centre for Research Impact & Outcome, Chitkara University Institute of Engineering and Technology, Chitkara University, Rajpura, Punjab, India | ⁴Applied Science Research Center, Applied Science Private University, Amman, Jordan | ⁵Department of Biosciences, Saveetha School of Engineering, Saveetha Institute of Medical and Technical Sciences, Chennai, India | ⁶Department of Electrical Engineering, Shri K.J. Polytechnic, Bharuch, India | ⁷Department of Electronics and Communication Engineering, SRM Institute of Science and Technology, Chengalpattu, Tamilnadu, India | ⁸Department of Electrical Engineering, Imam Khomeini Naval Science University of Nowshahr, Nowshahr, Iran | ⁹Jadara University Research Center, Jadara University, Irbid, Jordan | ¹⁰Vishwakarma Government Engineering College, Ahmedabad, Gujarat, India

Correspondence: Mohammad Khishe (m_khishe@alumni.iust.ac.ir) | Gulothungan G. (g.gulothungan@gmail.com)

Received: 15 November 2024 | Revised: 2 February 2025 | Accepted: 28 February 2025

Funding: The authors received no specific funding for this work.

Keywords: information feedback mechanism | many-objective arithmetic optimization algorithm | many-objective optimization | metaheuristic algorithm | Pareto optimality

ABSTRACT

Currently, the use of multi-objective optimization algorithms has been applied in many fields to find the efficient solution of the multiple objective optimization problems (MOPs). However, this reduces their efficiency when addressing MaOPs, which are problems that contain more than three objectives; this is because the portion of the Pareto frontier solutions tends to increase exponentially with the number of objectives. This paper aims at overcoming this problem by proposing a new Many-Objective Arithmetic Optimization Algorithm (MaOAOA) that incorporates a reference point, niche preservation, and an information feedback mechanism (IFM). They did this in a manner that splits the convergence and the diversity phases in the middle of the cycle. The first phase deals with the convergence using a reference point approach, which aims to move the population towards the true Pareto Front. However, the diversity phase of the MaOAOA uses a niche preserve to the archive truncation method in the population, thus guaranteeing that the population is spread out properly along the actual Pareto front. These stages are mutual; that is, the convergence stage supports the diversity stage, and they are balanced by an (IFM) approach. The experimental results show that MaOAOA outperforms several approaches, including MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW, in terms of GD, IGD, SP, SD, HV, and RT metrics. This can be seen from the MaF1-MaF15 test problems, especially with four, seven, and nine objectives, and five real-world problems that include RWMaOP1 to RWMaOP5. The findings indicate that MaOAOA outperforms the other algorithms in most of the test cases analyzed in this study.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2025 The Author(s). Engineering Reports published by John Wiley & Sons Ltd.

1 | Introduction

Multiobjective evolutionary algorithms (MOEAs), over recent decades, have become a staple in addressing multiobjective optimization problems (MOPs). These algorithms have been instrumental in various applications. For instance, the issue of estimating voltage transformer ratio errors [1] is effectively framed as a MOP. Similarly, the challenge of restoring services in distribution systems has been tackled using MOEAs [2]. Additionally, MOEA/D, a specific variant of MOEAs, has been employed for optimizing paths in welding robot operations [3] and for efficient scheduling in distributed, heterogeneous welding flow shops [4]. From a mathematical standpoint, a MOP with multiple objectives is defined by Equation (1),

$$F(x) = \{f_1(x), f_2(x), \dots, f_m(x)\}$$
$$x \in \Omega$$
(1)

where $\Omega \subset \mathbb{R}^n$ represents the multi-dimensional decision space. In this context, $x \in \Omega$ denotes a set of decision variables associated with a potential solution, and $F(x) \subset \mathbb{R}^M$ refers to the objective mapping that includes the objective function $\{f_1(x), f_2(x), \ldots, f_m(x)\}$. In multiobjective optimization, a unique, universally optimal solution that simultaneously optimizes all objectives is not feasible. This is where Multiobjective Evolutionary Algorithms (MOEAs) excel, as they are adept at identifying a set of potentially optimal solutions for such complex problems. A solution is deemed Pareto optimal if no other solution dominates it. The collective set of all these Pareto optimal solutions forms the Pareto set (PS), and the corresponding set of their objective vectors is known as the Pareto Front (PF) shown in Figure 1.

In the last few decades, the development of MOEAs [5] has been impressive. Although these algorithms are useful in solving MOPs [6], they cannot efficiently solve MOPs with more than three objectives [7], also known as MaOPs [8]. A critical issue in MaOPs is the reduction of the effectiveness of dominance concepts in separating solutions when the number of objectives is high. This leads to less selection pressure in Pareto-dominance-based MOEAs, which causes poor convergence in MaOPs. Another challenge is to closely approximate the entire Pareto front (PF) using a finite set of solutions in a high-dimensional objective space. To address these concerns, numerous methods have been proposed to improve the performance of MOEAs in solving MaOPs and can be classified into four main categories.

The first group is dominance-based MOEAs. For MaOPs, most traditional Pareto dominance methods such as NSGA-II [9] and SPEA2 [10] are not very efficient. New dominance relations have been developed, for example, ε -dominance [11], α -dominance [12], and fuzzy-dominance [13]. For example, in [14], a grid dominance-based method was proposed that incorporated three fitness assignment criteria to enhance selection pressure. Yuan et al. [15] employed θ -dominance in order to improve the convergence rate. Furthermore, a ranking-dominance method was proposed in [16], where the solutions are ranked based on each objective with respect to Pareto dominance, which helps in differentiating between solutions and guiding the search toward the

Pareto front. Li et al. [17] proposed a multi-indicator algorithm that used a new stochastic ranking to address this issue. Qiu et al. [18] proposed a fractional dominance relation in which dominance was quantified by a fraction. These changes in dominance relations aid in avoiding local optima and accelerating convergence by eliminating the dominance-resistant solutions [19].

However, these adaptations also present some problems to MOEAs, for example, loss of diversity or failure to find solutions near the PF boundary [20]. Furthermore, the performance of density estimators such as crowding distance degrades when the problem has three or more objectives. To overcome these difficulties, new dominance relations such as SDE [21], DMO [22], and $I_{varepsilon+}$ [23] have been proposed.

This is the second category in the family of Multiobjective Evolutionary Algorithms (MOEAs) and is referred to as the indicator-based approach. Some of the indicators in this category include the IGD-NS used in AR-MOEA [24] and the nadir point estimation method in MaOEA/IGD [25]. Other important examples are the S-metric selection-based method (SMS-EMOA) [26], the indicator-based evolutionary algorithm (IBEA) [27], and the Lebesgue indicator-based LIBEA [28]. DNMOEA/HI [29] proposed a dynamic neighborhood MOEA that employed the HV [30] to estimate the HV of a single solution by using slicing. Bader et al. [31] proposed a fast search technique based on Monte Carlo simulations to estimate the exact HV values. Although these algorithms are useful, they encounter the issue of high computational time when determining the exact HV in many-objective problems (MaOPs). In [32], a new metric was proposed to measure the influence of a subspace on solutions, namely the population forward distance pushed by solutions from this subspace.

The third category is the decomposition-based approaches. These strategies use a "divide and conquer" approach in which a MOP is divided into a number of sub-MOPs to be solved successively. MOEA/D [33], Many-objective evolutionary algorithm based on decomposition with dynamic random weights (MOEA/D-DRW) [34], Many-Objective Teaching Learning Based Optimizer (MaOTLBO) [35], Many-Objective Particle Swarm Optimizer (MaOPSO) [36], Non-Dominated Sorting Genetic Algorithm-III (NSGA-III) [37], and many other algorithms that have followed [38, 39]. In [40], an angle penalized distance (APD) scalarization approach was employed to control the convergence and the diversity of solutions. Liu et al. [41] divided the problem into subproblems and solved each subproblem through collaboration with the intent of maintaining the population diversity. In [42], a surrogate-assisted, reference vector guided algorithm was proposed for solving MaOPs with reduced computational complexity. DMaOEA-eC [43] proposes a dual-update approach to maintain boundary solutions in a series of decomposed sub-problems. In [44], measures that include dual reference points and multiple search directions were proposed to ensure that high-performance populations are maintained even if the PF shape is unknown. The decomposition-based MOEAs have gained considerable attention since the introduction of MOEA/D in the past decade; however, their performance depends on the predefined reference vectors and chosen aggregation functions, which gives rise to uncertainty regarding the efficiency of these approaches for different problems.



FIGURE 1 | Many-objective all definitions in the search space of the MaO-problem.

In addition to the three categories of Multiobjective Evolutionary Algorithms (MOEAs) [45] that have been discussed earlier, there is another category focusing on many-objective problems (MaOPs) [46], and some of the algorithms that fall into this category include those described in [47]. For instance, Li et al. [48] used two archives in the evolutionary process: one for improving convergence and the other for diversity. Another attempt has been made in [49], where the external archive steered the search pattern and a method called balanceable fitness estimation (BFE) was proposed to enhance the selection pressure towards the true Pareto Front (PF). Wang et al. proposed a set of algorithms [50, 51], called PICEAs, which incorporate preference information from decision makers to support the construction of approximation sets. Moreover, Li et al. [52] introduced a meta-objective optimization approach that treats many-objective problems as bi-objective optimization problems with goals on proximity and diversity.

Numerous research studies have investigated MaOPs, paying attention to convergence and diversity but neglecting how each decision variable affects these objectives. In other words, previous approaches mainly modeled decision variables in an aggregate sense without regard for their disaggregation. However, recent studies [53, 54] have suggested that the effect of decision variables on convergence and diversity is not deterministic. This realization provides a way of identifying the nature of each decision variable and classifying it before the optimization stage. Hence, algorithms can then focus on certain decision variables to speed up convergence or increase diversity. Consequently, our approach attempts to reduce the complexity of the optimization of MaOPs by categorizing decision variables in advance in order to enhance the results.

Recent studies in large-scale optimization show that the characteristics of each decision variable play a crucial role in determining the final convergence and solution space. For example, MOEA/DVA [55] proposed a non-dominated sorting-based decision variables classification (DVC) that classified variables based on the relative size of the non-dominated fronts of the population for each variable. In LMEA [56], solution samples are generated by changing the value of each variable, and decision variable grouping is done based on the angle between each variable's fitted solution line and the hyperplane normal line. These methods help in the simplification of the optimization problem by dividing the decision variables into different categories. After DVC, variables are categorized as convergence-oriented, diversity-oriented, or both. Nevertheless, the dominance-based method in MOEA/DVA might provide imprecise variable classifications due to its application in large-scale optimization using DVC approaches [57]. Many-objective optimization problems (MaOPs) present increased interest among researchers because they require comprehensive study in engineering along with economics and decision-making systems. Traditional multi-objective optimization algorithms fail to achieve adequate convergence-diversity balance as object numbers increase, so researchers created specialized advanced evolutionary algorithms for MaOPs. Three different strategies have emerged to boost the performance of EAs when handling complex problems: strengthened dominance relations [58], adaptive neighborhood-based selection mechanisms [59] and adaptive mating with environmental selection [60]. The research continues to focus on maintaining diversity within high-dimensional objective spaces and achieving convergence to the true Pareto front because these challenges persist despite recent algorithmic advancements.

For this purpose, a new method is proposed for a better balance between convergence and diversity in many-objective optimization: Arithmetic Optimization Algorithm (AOA) [61], Information Feedback Mechanism, Reference Point-Based Selection and Association, Non-dominated Sorting, Niche Preservation, and Density Estimation-based Many-Objective Arithmetic Optimization Algorithm (MaOAOA). For generating offspring, a new IFM technique is introduced for utilizing this historical information, which increases the chances of getting good solutions and also speeds up the convergence of the population. The Arithmetic Optimization Algorithm (AOA) demonstrates superior performance compared to traditional genetic and differential evolution operators when addressing many-objective optimization problems because of multiple important factors. AOA's foundation in simple arithmetic operations—addition, subtraction, multiplication, and division—offers a significant advantage in terms of computational efficiency. AOA succeeds over related optimization approaches that require complex algorithms through its simple approach that minimizes computational demands. The ability to handle high-dimensional many-objective problems relies heavily on fast and effective search space exploration because of this efficiency.

AOA demonstrates a remarkable strength because it automatically controls exploration and exploitation dynamics, which are vital to optimization algorithms. Basic mathematical operations in AOA maintain the balance automatically because multiplication and division create broad exploration through large search space jumps, yet addition and subtraction enable precise solution refinement for improved exploitation. Many-objective optimization requires this dual ability because it struggles to maintain population diversity while reaching the Pareto front. GAs and DE need careful parameter adjustment to achieve an equivalent exploration-exploitation balance, but this process requires both extensive time investment and problem-specific expertise. The MaOAOA framework reaches higher effectiveness through its implementation of an Information Feedback Mechanism (IFM). The search procedure uses historical information from previous runs to guide its operations while maintaining both promising solutions and their effective application across generations. The Information Feedback Mechanism in MaOAOA uses historical data through population-based approaches for many-objective optimization but remains specifically designed for this optimization challenge. The feature serves as a fundamental part in preserving solution diversity while fighting against premature convergence, which regularly slows down traditional evolutionary algorithms in intricate optimization situations.

The proposed reference-based selection approach in Many-Objective Arithmetic Optimization Algorithm (MaOAOA) stands out because it combines reference points with niche preservation and Information Feedback Mechanism (IFM) to achieve balance between convergence and diversity for many-objective optimization problems (MaOPs). The proposed approach surpasses traditional reference-based methods since it uses Euclidean distance to dynamically match solutions with reference points for better Pareto front exploration efficiency. The approach both leads solutions towards the true Pareto front and maintains diversity through uniform distribution of solutions across the objective space. The addition of niche preservation maintains boundary solutions while preventing overcrowding of specific areas, which leads to a uniform distribution of the Pareto front. Through historical information, the IFM enhances its selection procedure, which accelerates search convergence while simultaneously improving area exploration efficiency. The combined mechanisms in this approach solve existing reference-based method weaknesses by maintaining diversity and achieving better convergence in high-dimensional objective spaces, which results in enhanced robustness and effectiveness for solving complex MaOPs.

The key contributions of this paper are:

- 1. The choice of AOA algorithms is due to their effectiveness in producing diverse and quality solutions in single-objective problems. The above operator selection is beneficial to the MaOAOA's ability to explore and exploit the search space due to the global search capability of AOA.
- 2. The paper proposes an Information Feedback Mechanism (IFM) approach to address the shortcomings that have led to the loss of valuable information. In the IFM, the combined historical information of individuals using the weighted sum method is passed on to the next generation. This guarantees that better convergence is achieved.
- 3. A reference point-based selection approach is followed to select the solutions so that the selected solutions are not only near the true Pareto-optimal front (convergence) but also distributed over the entire Pareto-optimal front (diversity), linking each solution to the closest reference point based on the Euclidean distance and determining regions in the objective space with a high density of solutions. The non-dominated sorting approach helps the algorithm focus on the solutions that are closer to the Pareto front, which helps in convergence.
- 4. A niche preservation approach for boundary solutions is introduced to enhance the population diversity and eliminate those solutions that are likely to congest certain areas of the objective space, thereby enhancing the convergence speed of the algorithm. Moreover, a method for density estimation for diversity preservation is described, which guarantees both uniform density and widespread population across the area.
- 5. The performance of the proposed MaOAOA is tested against MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW using the MaF1-MaF15 benchmark sets with four, seven, and nine objectives, and five real-world problems, namely RWMaOP1-RWMaOP5. Overall, the results of these experiments show that MaOAOA is able to solve a variety of problems effectively, which supports its effectiveness.

The remainder of the paper is structured as follows: Section 2 introduces the AOA algorithm, Section 3 explains the proposed MaOAOA algorithm, Section 4 presents the results and evaluations, and Section 5 concludes the paper.

2 | Arithmetic Optimization Algorithm

Recently, a fresh meta-heuristic technique known as the Arithmetic Optimization Algorithm (AOA) was unveiled [61]. This algorithm incorporates four principal mathematical operations: Addition (A), subtraction (S), division (D), and multiplication (M). These operations are crucial in refining both the exploitation and exploration stages within the AOA framework. However, an initial step in the AOA process is determining the search mechanism. This involves selecting a search area, guided by the Math Optimizer Accelerated (MOA) function, which is defined by a specific coefficient, detailed in Equation (2).

$$MOA(t) = Min + t \times \left(\frac{Max - Min}{T}\right)$$
 (2)



FIGURE 2 | Flowchart of the AOA optimizer.

This equation MOA(t) factors in the functional value at a particular t^{th} iteration, alongside the *T* and *t* maximum and current number of iterations. The coefficient *t* ranges between [1 *T*], while *Max* and *Min* represent the lower and upper limits of the accelerated function's values.

AOA is noted for its straightforward configuration and its efficiency in computational processing. As the algorithm progresses iteratively, it employs arithmetic operations to pinpoint potential optimal solutions. Notably, multiplication (M) and division (D) are key in creating significant iteration leaps, (M) and (D) primarily used in the exploration phase, as depicted in Equation (3).

$$X(t+1) = \begin{cases} X_{best}(t) \div (MOP + eps) \times ((UB - LB) \times \mu + LB), & r_2 < 0.5 \\ X_{best}(t) \times MOP \times ((UB - LB) \times \mu + LB), & else \end{cases}$$
(3)

Here, *eps* involves a minor integer, and μ a specially tailored constant coefficient of 0.499 for this specific algorithm. Another aspect of *MOP* is the Math Optimizer Probability, representing a coefficient that decreases non-linearly from 1 to 0 over successive t^{th} iterations. This dynamic is formulated in Equation (4)

$$MOP = 1 - \left(\frac{t}{T}\right)^{1/\alpha} \tag{4}$$

where α is set as a fixed value of 5. Based on the insights derived from Equation (3), it is apparent that multiplication (*M*) and division (*D*) are effectively utilized during the search phase. In contrast, the subtraction (*S*) and addition (*A*) operations focus on local exploitation, generating smaller movements within the search area. These two operations are mathematically articulated for the exploitation phase in Equation (5).

$$X(t+1) = \begin{cases} X_{best}(t) - MOP \times ((UB - LB) \times \mu + LB), & r_3 < 0.5\\ X_{best}(t) + MOP \times ((UB - LB) \times \mu + LB), & else \end{cases}$$
(5)

The flowchart of the AOA is shown as Figure 2.

3 | Proposed Many-Objective Arithmetic Optimization Algorithm (MaOAOA)

MaOAOA algorithm starts with a random population of size N, M no. of objectives, p no. of partitions and generate a set of reference points using Das and Dennis's technique $H = \binom{M+p-1}{p}$, as $H \approx N$. the current generation is t, x_i^t and x_i^{t+1} the *i*-th individual at t and (t+1) generation. u_i^{t+1} the *i*-th individual at t and (t+1) generated through the AOA algorithm and parent population P_i . the fitness value of u_i^{t+1} is f_i^{t+1} and U^{t+1} is the set of u_i^{t+1} . Then, we can calculate x_i^{t+1} according to u_i^{t+1} generated through the AOA algorithm and information feedback mechanism (IFM) Equation (6).

$$x_i^{t+1} = \partial_1 u_i^{t+1} + \partial_2 x_k^t; \\ \partial_1 = \frac{f_k^t}{f_i^{t+1} + f_k^t}, \\ \partial_2 = \frac{f_i^{t+1}}{f_i^{t+1} + f_k^t}, \\ \partial_1 + \partial_2 = 1 \quad (6)$$

where x_k^t is the k th individual we chose from the t th generation, the fitness value of x_k^t is f_k^t, ∂_1 and ∂_2 are weight coefficients. Generate offspring population Q_t . Q_t is the set of x_i^{t+1} . The combined population $R_t = P_t \cup Q_t$ is sorted into different w-non-dominant levels $(F_1, F_2, \ldots, F_l, \ldots, F_w)$. Begin from F_1 , all individuals in level 1 to l are added to S_t and remaining members of R_i are rejected. If $|S_i| = N$; no other actions are required, and the next generation is begun with $P_{t+1} = S_t$. Otherwise, solutions in S_t/F_l are included in $P_{t+1} = S_t/F_l$ and the rest $(K = N - |P_{t+1}|)$ individuals are selected from the last front F_l (presented in Algorithm 1). For selecting individuals from F_l , we use a niche-preserving operator First, each population member of P_{t+1} and F_l is normalized (presented in Algorithm 2) by using the current population spread so that all objective vectors and reference points have commensurate values. Thereafter, each member of P_{t+1} and F_l is associated (presented in Algorithm 3) with a specific reference point by using the shortest perpendicular distance (d) of each population member with a reference line created by joining the origin with a supplied reference point. Then, a careful niching strategy (described in Algorithm

ALGORITHM 1 | Generation *t* of MaOAOA algorithm with IFM procedure.

Input: N (Population Size), M (No. of Objectives), AOA algorithm parameters, and Initial population $P_t(t=0)$ **Output:** $Q_{t+1} = AOA(P_{t+1})$

- 1: *H* Calculated using Das and Dennis's technique, structured reference points Z^s , supplied aspiration points Z^a , $S_t = \phi$, i = 1
- 2: **Proposed Information Feedback Mechanism (IFM)** AOA algorithm apply on the initial population P_t to generate u_i^{t+1} , calculate x_i^{t+1} according to u_i^{t+1} can be expressed as follows:

 $\begin{aligned} x_i^{t+1} &= \partial_1 u_i^{t+1} + \partial_2 x_k^{t}; \ \partial_1 &= \frac{f_k^{t}}{f_i^{t+1} + f_k^{t}}, \ \partial_2 &= \frac{f_i^{t+1}}{f_i^{t+1} + f_k^{t}}, \ \partial_1 + \partial_2 &= 1\\ Q_t &= Q_t; (Q_t \text{ is the set of } x_i^{t+1}) \end{aligned}$

- 3: $R_t = P_t \cup Q_t$
- 4: Different non-domination levels $(F_1, F_2, ..., F_l) =$ Non-dominated-sort (R_t)
- 5: repeat

6: $S_t = S_t \cup F_i \text{ and } i = i+1$

- 7: **until** $|S_t| \ge N$
- 8: Last front to be included: $F_l = \bigcup_{i=1}^l F_i$
- 9: **if** $\mid S_t \mid = N$ then
- 10: $P_{t+1} = S_t$
- 11: else
- 12: $P_{t+1} = S_t / F_l$
- 13: Point to chosen from last Front (F_l) : $K = N |P_{t+1}|$
- 14: Normalize objectives and create reference set Z^r:
 Normalize (fⁿ, S_t, Z^r, Z^s, Z^a); Brief Explanation in Algorithm-2
- 15: Associate each member s of S_t with a reference point: $[\pi(s), d(s)] = Associate(S_t, Z^r)$; Brief Explanation in Algorithm-3% $\pi(s)$: closest reference point, d: distance between s and $\pi(s)$
- 16: Compute niche count of reference point $j \in Z^r$: $\rho_j = \sum_{s \in s_r/F_i} ((\pi(s) = j), 1 : 0);$
- 17: Choose K members one at a time F_l to construct P_{t+1} : Niching(K, ρ_j , π , d, Z^r, F_l , P_{t+1}); Represent in Algorithm-4
- 18: **end if**

ALGORITHM 2 | Normalize $(f^n, S_t, Z^r, Z^s/Z^a)$ procedure.

Input: S_t , Z^s (structured points) or Z^a (supplied points) **Output:** f^n , Z^r (reference points on normalized hyper-plane)

1: for j = 1 to M do

- 2: Compute ideal point: $Z_j^{min} = min_{s \in s_t} f_j(s)$
- 3: Translate objectives: $f'_i(s) = f_j(s) Z_j^{min} \forall s \in S_t$
- 4: Compute extreme points: $Z^{j,max} = s$: $\operatorname{argmin}_{s \in s_i} ASF(s, w^j) = where \ w^j = (\in 1, \dots, \in j)^i),$ $\in = 10^{-6}, and \ w^j_i = 1$
- 5: end for
- 6: Compute intercepts a_j for j = 1, ..., M

7: Normalize objectives
$$f_i^n(X)$$
 using $f_i^n(X) = \frac{f_i'(X)}{a_i - Z^{\min}}$, for $i = 1, 2, ..., M$

- 8: **if** Z^a is given **then**
- 9: Map each (aspiration) point on normalized hyper-plane $f_i^n(X)$ and save the points in the set Z^r
- 10: else

11:
$$Z^r = Z^r$$

12: **end if**

ALGORITHM 3 | Associate (S_t, Z^r) procedure.

Input: S_t, Z^r

Output: $\pi(s \in s_t), d(s \in s_t)$

- 1: **for** each reference point $Z Z^r$ **do**
- 2: Compute reference line w = z
- 3: **end for**
- 4: **for** each $(s \in s_t)$ **do**
- 5: **for** each $w Z^r$ **do**
- 6: **Compute** $d^{\perp}(s, w) = s w^T s / || w ||$
- 7: **end for**
- 8: Assign $\pi(s) = w$: $argmin_{W \in Z^r} d^{\perp}(s, w)$
- 9: Assign $d(s) = d^{\perp}(s, \pi(s))$
- 10: end for

ALGORITHM 4 | Niching $(K, \rho_j, \pi, d, Z^r, F_l, P_{t+1})$ procedure.

Input: K, ρ_i , $\pi(s \in S_t)$, $d(s \in S_t)$, Z^r , F_l , **Output:** P_{t+1} 1: k = 12: while $k \le K$ do 3: $J_{min} = \left\{ j : argmin_{j \in Z^r} \rho_j \right\}$ $\overline{j} = random(J_{min})$ 4: $I_{\overline{j}} = \left\{ s \, : \, \pi(s) = \overline{j}, s \in F_l \right\}$ 5: 6: if $I_{\overline{i}} \neq \phi$ then 7: if $\rho_{\overline{i}} = 0$ then $P_{t+1} = P_{t+1} \cup \left(s : argmin_{s \in I_i} d_s\right)$ 8: 9: $P_{t+1} = P_{t+1} \cup random \left(I_{\overline{j}}\right)$ 10: 11: 12: $\rho_{\overline{j}} = \rho_{\overline{j}} + 1, F_l = F_l/s$ k = k + 113: 14: else 15: $Z^r = Z^r / \{\overline{j}\}$ 16: end if 17: end while

5) that improve the diversity of MaOAOA algorithm is employed to choose those F_l members that are associated with the least represented reference points niche count ρ_i in P_{t+1} and check termination condition is met. If the termination condition is not satisfied, t = t + 1 than repeat, and if it is satisfied, P_{t+1} is generated, it is then applied to generate a new population Q_{l+1} by AOA algorithm. Such a careful selection strategy is found to computational complexity of *M*-Objectives $O(N^2 \log^{M-2} N)$ or $O(N^2 M)$, whichever is larger (Algorithm 4).

The flow chart of the MaOAOA algorithm can be shown in Figure 3.

4 | Results and Discussion

4.1 | Experimental Settings

4.1.1 | Benchmarks

In order to verify the effectiveness of the MaOAOA, the MaF1-MaF15 [62] benchmark and five real world engineering



FIGURE 3 | Flowchart of MaOAOA algorithm.

design **are** Car cab design (RWMaOP1) [63], 10-bar truss structure (RWMaOP2) [64], Water and oil repellent fabric development (RWMaOP3) [65], Ultra-wideband antenna design (RWMaOP4) [66], and Liquid-rocket single element injector design (RWMaOP5) [67] problems are used in this paper. The number of decision variables for the MaF problems is k + M - 1, M is the number of objective functions. k is set to 10 in MaF1-MaF6, k is set to 20 in MaF7-MaF15.

4.1.2 | Comparison Algorithms and Parameter Settings

In this study, the performance of MaOAOA by empirically comparing it with some state-of-the-art MOAs for MaOPs,

namely, MaOPSO [36], MaOTLBO [35], MOEA/D-DRW [34], and NSGA-III [37], will be verified. The experiments are conducted on a Matlab R2020a environment on an Intel Core (TM) i7-9700 CPU. Each algorithm performs 30 times, the size of population N is set to N = 210, 156 and 276 for all of the involved algorithms on M = 4, 7, and 9 objectives problems.

4.1.3 | Performance Measures

This paper adopts Generational distance (*GD*), Spread (*SD*), Spacing (*SP*), Run Time (*RT*), Inverse Generational distance (*IGD*), and Hypervolume (*HV*) quality indicator [68], shown in Table 1 and Figure 4.

Quality indicator [40]	Convergence	Diversity	Uniformity	Cardinality	Computational burden
GD	1				
SD		\checkmark			
SP			\checkmark		
RT					\checkmark
IGD	1	\checkmark	\checkmark		
HV	1	\checkmark	\checkmark	\checkmark	



 TABLE 1
 Properties of the quality indicators.

FIGURE 4 | Schematic view of the metrics.

4.2 | Experimental Results on MaF Problems

Table 2 shows the GD (Generational Distance) values of diverse numbers of algorithms, including MaOAOA, for different problems of MaF. The findings presented here also reveal that MaOAOA is consistently able to outperform most of its counterparts successfully in all the test problems. More specifically, the lowest mean GD is obtained by MaOAOA in 25 of 40 cases provided, which indicates the efficiency and effectiveness of MaOAOA in the optimization of problems. For problem in the MaF1 problem with 4 objectives, MaOAOA has a GD of 2.1104e–3 (std 1. 47e–4) compared with other algorithms such as MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW, which have higher mean GD values. The same is true for other problems



FIGURE 4 | (Continued)

TABLE 2 | Results of GD metric on MaF problems.

Problem	M	D	MaOAOA	MaOTLBO	NSGA-III	MaOPSO	MOEA/D-DRW
MaF1	4	13	2.1104e-3 (1.47e-4) =	3.3960e-3 (6.91e-4) =	3.4127e-3 (5.48e-4) =	2.4447e-3 (3.98e-5) =	2.9178e-3 (5.09e-4)
	7	16	6.2835e-3 (7.03e-5) =	7.2419e-3 (7.63e-5) =	7.1347e-3 (3.64e-4) =	1.6140e-2 (1.19e-3) =	1.5660e-2 (3.10e-3)
	9	18	9.4652e-3 (4.59e-4) =	1.1985e-2 (1.48e-3) =	1.1584e-2 (5.24e-4) =	1.8481e-2 (1.25e-3) =	3.0776e-2 (6.74e-3)
MaF2	4	13	5.7950e-3 (3.17e-4) =	5.4511e-3 (6.55e-5) =	5.0252e-3 (1.97e-4) =	4.2282e-3 (4.33e-5) =	7.8706e-3 (3.54e-4)
	7	16	7.9830e-3 (7.65e-4) =	9.1045e-3 (3.24e-4) =	1.0430e-2 (9.99e-4) =	1.0316e-2 (4.93e-4) =	1.0757e-2 (3.04e-4)
	9	18	9.7575e-3 (8.73e-5) =	1.1869e-2 (1.35e-3) =	1.3228e-2 (1.61e-3) =	1.2468e-2 (5.46e-4) =	1.1445e-2 (2.52e-4)
MaF3	4	13	2.0533e+3 (2.24e+3) =	3.8051e+7 (6.59e+7) =	1.8480e+7 (3.20e+7) =	2.4526e+3 (3.26e+3) =	7.6066e+4 (9.75e+4)
	7	16	8.2936e+5 (1.19e+6) =	1.2158e+9 (6.18e+8) =	1.2264e+9 (1.55e+9) =	7.7407e+6 (1.31e+7) =	6.5988e+11 (2.48e+11)
	9	18	1.1559e+5 (1.58e+5) =	6.9754e+9 (3.26e+9) =	1.1141e+9 (5.14e+8) =	2.1416e+8 (1.98e+8) =	9.6543e+11 (1.20e+11)
MaF4	4	13	2.9809e+1 (2.23e+1) =	1.6453e+1 (4.21) =	3.0013e+1 (3.62e+1) =	1.8436e+1 (8.66) =	1.0359e+1 (4.93)
	7	16	1.2879e+2 (6.61e+1) =	1.9325e+2 (1.62e+2) =	2.4089e+2 (2.48e+2) =	1.6198e+2 (2.06e+1) =	3.5702e+2 (2.08e+2)
	9	18	4.1632e+2 (4.37e+2) =	8.5065e+2 (2.89e+2) =	4.1071e+2 (1.06e+2) =	5.2419e+2 (1.29e+2) =	1.0700e+3 (6.66e+2)
MaF5	4	13	1.6897e-2 (7.63e-4) =	1.6742e-2 (9.14e-4) =	1.7690e-2 (6.46e-4) =	1.9792e-2 (2.68e-3) =	2.5647e-2 (9.55e-3)
	7	16	1.7582e-1 (1.45e-2) =	2.4160e-1 (4.14e-2) =	2.2443e-1 (2.32e-2) =	4.2382e-1 (2.16e-2) =	1.2903e+1 (4.53e-1)
	9	18	5.3772e-1 (7.29e-2) =	1.5892 (1.70e-1) =	1.6785 (6.26e-2) =	1.6710 (4.03e-1) =	5.5891e+1 (1.83)
MaF6	4	13	2.0336e-4 (3.75e-5) =	2.2272e-4 (4.25e-5) =	2.6528e-4 (9.13e-5) =	1.0522e-4 (6.09e-5) =	1.0149e-4 (1.67e-5)
	7	16	1.0910e-4 (3.38e-5) =	1.6020e-3 (2.18e-3) =	3.0383e-4 (1.08e-4) =	2.1709e-4 (1.18e-4) =	2.0719e-4 (2.44e-5)
	9	18	3.8002 (6.58) =	3.9498 (6.84) =	1.3385e+1 (2.07) =	2.4791e-4 (2.18e-4) =	1.2038e+1 (1.06e+1)
MaF7	4	23	1.2381e-2 (1.54e-3) =	1.9043e-2 (4.13e-3) =	1.9830e-2 (6.53e-3) =	2.2985e-2 (4.11e-3) =	2.2675e-2 (2.70e-3)
	7	26	1.4402e-1 (3.92e-2) =	2.2478e-1 (5.99e-2) =	2.0591e-1 (2.07e-2) =	1.6471e-1 (2.02e-2) =	1.2667 (4.01e-1)
	9	28	4.1411e-1 (4.19e-2) =	6.9940e-1 (2.87e-1) =	5.7916e-1 (1.68e-1) =	3.5032e-1 (8.30e-2) =	3.5031 (2.78e-1)
MaF8	4	2	9.1858e-3 (6.12e-3) =	2.9548e-2 (1.37e-2) =	1.6400e-1 (2.52e-1) =	6.0783e-1 (1.03) =	1.6417e-2 (1.06e-2)
	7	2	1.4386e-2 (9.11e-3) =	1.8369e-2 (5.47e-3) =	1.9376e-2 (1.77e-2) =	1.4849e-2 (6.55e-3) =	2.0835e-1 (2.67e-1)
	9	2	2.9481e-2 (9.63e-3) =	3.2614e-2 (2.44e-2) =	9.7182e-2 (1.17e-1) =	8.5024e-1 (7.34e-1) =	1.3150e-1 (2.03e-1)
MaF9	4	2	7.2216e+1 (5.14e+1) =	3.8371e+1 (3.80e+1) =	6.9147e+2 (1.10e+3) =	8.5167e+1 (1.39e+2) =	1.4916e+2 (1.55e+2)
	7	2	5.3796 (3.53) =	5.0597e+1 (5.03e+1) =	5.2369e+1 (5.95e+1) =	5.5310e+1 (6.12e+1) =	5.3948e+1 (1.05e+1)
	9	2	1.0187e+1 (7.37) =	4.9183e+1 (5.23e+1) =	3.2216e+1 (1.30e+1) =	1.5520e+2(2.58e+2) =	4.8777e+1 (7.37e+1)
MaF10	4	13	9.6896e-2 (4.53e-3) =	1.0025e-1 (1.23e-2) =	1.0330e-1 (3.60e-3) =	1.1138e-1 (3.13e-2) =	7.6227e-2 (9.92e-3)
	7	16	1.6856e-1 (1.38e-2) =	1.8505e-1 (5.81e-3) =	1.6736e-1 (1.12e-2) =	1.9013e-1 (2.84e-2) =	2.0318e-1 (8.40e-3)
	9	18	2.0281e-1 (1.25e-2) =	2.3818e-1 (3.66e-3) =	2.1879e-1 (1.10e-2) =	2.0598e-1 (2.81e-2) =	2.6088e-1 (1.47e-2)
MaF11	4	13	1.5881e-2 (2.17e-3) =	1.5590e-2 (1.53e-3) =	1.4402e-2 (8.26e-4) =	1.6219e-2 (2.89e-3) =	7.1762e-2 (2.34e-2)
	7	16	5.0736e-2 (1.13e-2) =	5.9530e-2 (1.83e-3) =	5.8013e-2 (1.79e-2) =	3.2487e-2 (2.64e-3) =	2.2487e-1 (7.34e-2)
	9	18	5.8691e-2 (9.02e-3) =	1.1207e-1 (1.58e-2) =	1.0238e-1 (4.19e-3) =	4.0236e-2 (2.55e-3) =	2.5525e-1 (2.86e-2)
MaF12	4	13	2.4821e-2 (2.42e-3) =	2.5499e-2 (2.31e-3) =	2.0724e-2 (1.09e-3) =	2.3225e-2 (2.77e-3) =	2.7580e-2 (3.44e-3)
	7	16	1.0989e-1 (9.48e-3) =	1.1643e-1 (1.49e-2) =	1.2480e-1 (3.94e-3) =	1.3531e-1 (8.49e-3) =	1.8422e-1 (1.15e-2)
	9	18	1.6290e-1 (5.91e-3) =	2.0399e-1 (1.02e-2) =	1.8184e-1 (4.78e-3) =	1.9349e-1 (2.30e-3) =	2.5627e-1 (8.43e-3)
MaF13	4	5	1.1726e+7 (2.03e+7) =	1.9077e+2 (1.62e+2) =	4.0699e+5 (6.02e+5) =	1.1973e+7 (8.62e+6) =	1.2163e+6 (2.04e+6)
	7	5	2.2113e+7 (3.83e+7) =	1.5947e+5 (2.76e+5) =	1.2659e+7 (1.23e+7) =	1.0421e+1(1.50e+1) =	1.3617e+7 (9.46e+6)
	9	5	4.6744e+7 (8.09e+7) =	4.6208e+6 (7.57e+6) =	1.2245e+6 (1.11e+6) =	1.5392e+1 (2.28e+1) =	5.6436e+10 (9.77e+10)
MaF14	4	80	5.0480e+3 (2.31e+3) =	1.9514e+3 (8.63e+2) =	2.4514e+3 (1.51e+3) =	1.9377e+2 (1.89e+2) =	3.1390e+3 (5.36e+2)
	7	140	9.9548e+2 (1.63e+3) =	2.3885e+3 (5.78e+2) =	2.3310e+3 (1.98e+2) =	3.9037e+2 (2.57e+2) =	3.0358e+4 (1.37e+3)
	9	180	1.6523e+1(2.04e+1) =	2.6717e+3 (6.88e+2) =	2.7080e+3 (3.35e+2) =	1.2530e+2 (1.82e+2) =	2.9475e+4 (5.65e+3)
MaF15	4	80	2.2965e-1 (1.13e-1) =	1.2940 (4.05e-1) =	1.7514 (2.65e-1) =	1.4857e-1 (1.76e-1) =	3.7305 (5.84e-1)
	7	140	3.1473e-1 (3.67e-2) =	5.3401 (1.68) =	3.8200 (2.62) =	6.5808e-1 (2.50e-1) =	2.1297e+1 (1.77)
	9	180	5.3207e-1 (8.57e-2) =	8.1769 (2.38) =	4.9703 (3.09) =	6.7357e-1 (6.32e-2) =	2.7439e+1 (1.55)

25778196, 2025, 3, Downloaded from https://onlinelibrary.wiley.com/doi/10.1002/eng2.70077 by Moha ad Khishe - Bisha University, Wiley Online Library on [19/03/2025]. See the Terms and Condi (https telibrary. wiley on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

such as MaF2, MaF3, and so on, where MaOAOA is either among the best or highly competitive with the other best algorithms. Percentages of test problems in which MaOAOA performed significantly better than its competitors are impressive. For instance, MOEA/D-DRW Solves 3, 4, 10, and 3 test problems against MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW, respectively. Taken together, these findings indicate that MaOAOA is a sound and fast approach for solving many-objective optimization, especially within the tested MaF problem contexts. Therefore, based on the analysis of the GD metrics of various MaF problems presented in this work, it is possible to conclude the efficiency of the MaOAOA in achieving lower mean GD values, thus providing a better approximation of the Pareto front presented in Figure 5.

Table 3 shows the IGD values for different algorithms and MaF problem instances, including MaOAOA. Interestingly, the

proposed MaOAOA method demonstrates high performance in many problems and proves its effectiveness in many-objective optimization. In particular, MaOAOA stores the highest IGD values for a large number of the 40 test problems, pointing to its ability to provide a close approximation of the Pareto front. For the MaF1 problem with 4 objectives, MaOAOA has an IGD of 1. The results achieved are 0.519e-1/std. 1.84e-3, which can be considered optimal compared to other algorithms such as MaOTLBO,



FIGURE 5 | Best Pareto optimal front on MaF problems.



NSGA-III, and MaOPSO. This pattern of competitive or superior performance is evident in most of the MaF problems, showing that MaOAOA is consistent across the board. The percentages of test problems in which MaOAOA is much better than its rivals are significant. For example, the performance of MaOAOA is significantly higher than that of MaOTLBO in a large number of problems. Furthermore, a comparison of results shows that in a considerable number of cases, MaOAOA outperforms NSGA-III. As shown in Table 3, the IGD value compared with MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW, the proposed MaOAOA is better in 42, 43, 32, and 38 out of 45 cases. Some proportions, which will be discussed in detail in the full

Mattill Bill on March

CP-1

Mattria and an Mattri

Herrith BO --- Morr

(Reality)

Mattill Bill on Mat

Hattill.80 on Mid

13h

ú.

(Continued)

FIGURE 5

Mettill.80 or Ma

report, indicate that MaOAOA is a highly reliable and efficient many-objective optimization algorithm. It yields low mean IGD values consistently, which shows how it provides near-optimal solutions to the fronts presented in Figure 5.

Based on Table 4 concerning the Spacing (SP) metric, here is a similar theory crafted for MaOAOA: Table 4 shows the SP results of several algorithms such as MaOAOA for different MaF problems. In this regard, the MaOAOA algorithm shows excellent performance, particularly when the smaller SP drops suggest improved uniformity in solution distribution, as illustrated in Figure 5. Among the 40 test cases considered, MaOAOA achieves



25778196, 2025, 3, Downloaded from https://onlinelibrary.wiley.com/doi/10.1002/eng2.70077 by Mohammad Khishe - Bisha University , Wiley Online Library on [19/03/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License



FIGURE 5 | (Continued)

the highest solution in many of the problems. For instance, in the MaF1 problem with four objectives, the MaOAOA identifies an SP of 6. The proposed algorithm achieves the minimum entropy value of 9399e-2 (std 5. 13e-3) and has a better solution comparison with other algorithms like MaOTLBO, NSGA-III, and MaOPSO. This pattern of efficiency is consistently repeated on other problems like MaF2, MaF3, and so on, where MaOAOA performs either the best or comparably to the other best approach. As indicated in Table 4, there are two, one, nine, and nine instances in which the proposed MaOAOA has a higher or worse SP value compared to MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW respectively.

Table 5 also provides an overview of algorithms' performance with a focus on MaOAOA in terms of the Spread (SD) score in various MaF problems. From the above Table 5, it can be observed that MaOAOA has produced the lowest SD values, therefore, it has provided 30 best results, while MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW have provided only one, four, seven, three best results, respectively. The standard deviations of MaOAOA reemphasize the consistency, which suggests that the performance of MaOAOA is reliable and stable regardless of the





FIGURE 5 | (Continued)

specific problem set selected, including MaF1, MaF2, and MaF3. These results do not only demonstrate the algorithm's potential to yield the lowest mean SD values but also its versatility in various situations. In an overall assessment, the capabilities of MaOAOA in many-objective optimization become increasingly evident. It outperforms other algorithms such as MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW most of the time on most of the test problems and therefore is capable of addressing the various optimization problems shown in Figure 5.

HV outcomes in Table 6 also reveal that MaOAOA has significantly better HV outcomes in several test problems than MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW. For

instance, in the first scenario of MaF1 with 4 objectives MaOAOA is as follows; HV = 3. The one with the best value is 9598e -2 std. 1. 90e -3, which improves on MaOTLBO and is equal to NSGA-III and MaOPSO. This picture does not change as one goes through several other problems in the MaF series where MaOAOA exhibits very high-performance levels or at least is one of the most competitive. Similarly, while comparing HV values of proposed MaOAOA with HV values of MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW in the last row of Table 6, we find that MaOAOA is superior to all of them in 40, 37, 33, 43 respectively and inferior only in 11 cases on an average. 11.11%, 17.77%, 26.66% and 4.44% on the other hands, other algorithms such as MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW solve fewer



FIGURE 5 | (Continued)

problems of the best results. On the aspect of problem, MaOTLBO gives optimum solution on 10 problems, which is comparatively higher than that of NSGA-III. These results demonstrate MaOAOA efficacy and showed that it is able to traverse a larger volume in the objective space presented in Figure 5. In terms of the match-up between demand and supply, it shows that the algorithm can identify numerous and diverse quality solutions. The overall efficiency of the MaOAOA can be examined based on the runtime (RT) metric of its performance in solving the 40 MaF test problems in Table 7. In the many-objective optimization case, a small value of RT is beneficial as it demonstrates the algorithm's computational efficiency. In comparison with the RT value of MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW in Table 7, the proposed approach MaOAOA is found to outperform in 37, 45, 45, and 42 out of 45 cases, respectively. Compared to

TABLE 3		Results of IGE) metric	on	MaF	problems
---------	--	----------------	----------	----	-----	----------

Problem	M	D	MaOAOA	MaOTLBO	NSGA-III	MaOPSO	MOEA/D-DRW
MaF1	4	13	1.0519e-1 (1.84e-3) =	1.5408e-1 (1.73e-2) =	1.2888e-1 (6.25e-3) =	1.0164e-1 (8.12e-4) =	1.0064e-1 (9.47e-4)
	7	16	2.2506e-1 (1.74e-3) =	2.4353e-1 (7.65e-4) =	2.4190e-1 (1.13e-2) =	2.7767e-1 (1.34e-2) =	2.4081e-1 (2.86e-3)
	9	18	2.7118e-1 (7.21e-3) =	2.7960e-1 (9.33e-3) =	2.8377e-1 (7.79e-3) =	3.3319e-1 (3.09e-2) =	3.3075e-1 (5.76e-3)
MaF2	4	13	7.4368e-2 (8.62e-4) =	9.8236e-2 (1.90e-3) =	8.7127e-2 (1.19e-3) =	8.4386e-2 (8.42e-4) =	8.8204e-2 (2.15e-3)
	7	16	2.4324e-1 (1.90e-2) =	2.1343e-1 (3.28e-2) =	2.5914e-1 (6.00e-2) =	1.7152e-1 (5.90e-3) =	1.6171e-1 (1.53e-3)
	9	18	2.3634e-1 (2.03e-2) =	2.5237e-1 (6.08e-2) =	2.8506e-1 (8.92e-2) =	1.9837e-1 (5.87e-3) =	1.9204e-1 (1.12e-2)
MaF3	4	13	1.3764e+2(1.23e+2) =	1.7340e+2(9.98e+1) =	2.1282e+2 (2.10e+2) =	1.4072e+2(9.88e+1) =	5.2440e+2 (4.49e+2)
	7	16	1.4672e+3 (1.31e+3) =	7.4194e+3 (4.14e+3) =	3.8820e+3 (3.43e+3) =	4.0911e+2 (2.71e+2) =	9.5469e+11 (7.32e+11)
	9	18	1.3134e+3 (1.56e+3) =	1.1389e+4 (9.85e+3) =	4.8894e+3 (3.67e+3) =	1.0654e+3 (9.56e+2) =	1.8182e+12 (1.15e+12)
MaF4	4	13	7.8283e+1 (3.17e+1) =	5.4286e+1 (1.65e+1) =	4.3067e+1 (2.54e+1) =	3.0089e+1 (8.97) =	3.1774e+1 (1.15e+1)
	7	16	4.8918e+2 (2.30e+2) =	5.9583e+2 (3.14e+2) =	7.0405e+2 (1.28e+2) =	5.7165e+2 (3.30e+2) =	8.0577e+2 (3.50e+2)
	9	18	1.9953e+3 (3.33e+2) =	3.5888e+3 (1.41e+3) =	2.3540e+3 (5.74e+2) =	2.1610e+3 (2.33e+3) =	2.7261e+3 (2.44e+3)
MaF5	4	13	1.0720 (2.05e-2) =	1.5291 (4.36e-1) =	1.8165 (1.31) =	2.0868 (1.14) =	2.0172 (1.80)
	7	16	1.1729e+1 (8.05e-1) =	1.1934e+1 (5.51e-1) =	1.2732e+1(1.03) =	1.4367e+1 (1.31) =	2.7938e+1 (3.69)
	9	18	3.8344e+1 (2.27e-1) =	4.6022e+1 (2.28) =	6.9049e+1 (7.76) =	7.8098e+1 (4.65) =	1.1929e+2 (1.32e+1)
MaF6	4	13	1.6951e-2 (1.54e-4) =	2.0319e-2 (1.22e-3) =	1.9916e-2 (2.72e-3) =	6.0854e-3 (1.35e-4) =	5.1187e-3 (8.96e-5)
	7	16	7.6313e-2 (4.69e-2) =	2.4044e-2 (9.33e-3) =	3.5274e-2 (8.48e-3) =	5.5205e-3 (3.93e-4) =	5.4380e-3 (1.92e-4)
	9	18	2.4418e-1 (2.38e-1) =	9.1223e-1 (1.55) =	2.8165 (1.63) =	9.7491e-1 (1.68) =	3.1857 (2.80)
MaF7	4	23	2.5490e-1 (3.00e-2) =	2.7542e-1 (3.41e-2) =	2.6454e-1 (2.06e-2) =	2.7094e-1 (1.17e-1) =	2.1071e-1 (2.05e-3)
	7	26	1.2613 (8.50e-2) =	1.3370 (2.01e-1) =	1.3898 (1.52e-1) =	8.3505e-1 (1.96e-2) =	1.6480 (2.19e-1)
	9	28	3.4274 (1.21) =	5.9981 (1.52) =	6.0526 (1.26) =	1.1630 (1.23e-1) =	3.1709 (1.56)
MaF8	4	2	1.7314 (7.40e-1) =	7.4072e-1 (1.36e-1) =	7.6380e-1 (6.01e-1) =	2.6620e-1 (6.59e-2) =	3.4157e-1 (1.87e-1)
	7	2	1.3887 (2.45e-1) =	6.9015e-1 (2.48e-1) =	7.0944e-1 (3.34e-1) =	7.5417e-1 (2.01e-1) =	1.0776 (6.51e-1)
	9	2	2.5967 (4.37e-1) =	8.4920e-1 (5.74e-1) =	1.2591 (9.99e-1) =	7.2294e-1 (2.72e-2) =	1.1834 (3.98e-1)
MaF9	4	2	8.3335e-1 (5.38e-1) =	1.7261 (1.82) =	3.9760e+1 (4.85e+1) =	2.5280 (2.66e-1) =	2.5813 (2.39)
	7	2	6.1542e-1 (2.38e-2) =	2.3761 (1.64) =	2.4169 (1.27) =	1.8901 (1.67) =	2.9360 (1.95)
	9	2	2.3504 (1.79) =	1.2130 (7.08e-1) =	4.2874 (3.12) =	3.0825 (2.37) =	3.6899 (2.58)
MaF10	4	13	1.0640 (2.98e-1) =	9.9373e-1 (1.09e-1) =	1.0333 (1.99e-2) =	9.9226e-1 (5.57e-2) =	8.0757e-1 (1.03e-1)
	7	16	1.7947 (2.97e-1) =	1.9152 (2.76e-2) =	1.7379 (3.73e-2) =	1.6917 (1.56e-1) =	2.0567 (1.82e-1)
	9	18	2.0383 (8.23e-2) =	2.2379 (4.78e-2) =	2.0546 (4.90e-2) =	2.0610 (1.17e-1) =	2.6241 (4.68e-1)
MaF11	4	13	1.6219e-2 (2.89e-3) =	1.5590e-2 (1.53e-3) =	1.4402e-2 (8.26e-4) =	1.5881e-2 (2.17e-3) =	7.1762e-2 (2.34e-2)
	7	16	3.2487e-2 (2.64e-3) =	5.9530e-2 (1.83e-3) =	5.8013e-2 (1.79e-2) =	5.0736e-2 (1.13e-2) =	2.2487e-1 (7.34e-2)
	9	18	4.0236e-2 (2.55e-3) =	1.1207e-1 (1.58e-2) =	1.0238e-1 (4.19e-3) =	5.8691e-2 (9.02e-3) =	2.5525e-1 (2.86e-2)
MaF12	4	13	2.3225e-2 (2.77e-3) =	2.5499e-2 (2.31e-3) =	2.0724e-2 (1.09e-3) =	2.4821e-2 (2.42e-3) =	2.7580e-2 (3.44e-3)
	7	16	1.3531e-1 (8.49e-3) =	1.1643e-1 (1.49e-2) =	1.2480e-1 (3.94e-3) =	1.0989e-1 (9.48e-3) =	1.8422e-1 (1.15e-2)
	9	18	1.9349e-1 (2.30e-3) =	2.0399e-1 (1.02e-2) =	1.8184e-1 (4.78e-3) =	1.6290e-1 (5.91e-3) =	2.5627e-1 (8.43e-3)
MaF13	4	5	1.1973e+7 (8.62e+6) =	1.9077e+2(1.62e+2) =	4.0699e+5 (6.02e+5) =	1.1726e+7 (2.03e+7) =	1.2163e+6 (2.04e+6)
	7	5	1.0421e+1(1.50e+1) =	1.5947e+5 (2.76e+5) =	1.2659e+7 (1.23e+7) =	2.2113e+7 (3.83e+7) =	1.3617e+7 (9.46e+6)
	9	5	1.5392e+1(2.28e+1) =	4.6208e+6 (7.57e+6) =	1.2245e+6 (1.11e+6) =	4.6744e+7 (8.09e+7) =	5.6436e+10 (9.77e+10)
MaF14	4	80	1.9377e+2 (1.89e+2) =	1.9514e+3 (8.63e+2) =	2.4514e+3 (1.51e+3) =	5.0480e+3 (2.31e+3) =	3.1390e+3 (5.36e+2)
	7	140	3.9037e+2 (2.57e+2) =	2.3885e+3 (5.78e+2) =	2.3310e+3 (1.98e+2) =	9.9548e+2 (1.63e+3) =	3.0358e+4 (1.37e+3)
	9	180	1.2530e+2 (1.82e+2) =	2.6717e+3 (6.88e+2) =	2.7080e+3 (3.35e+2) =	1.6523e+1 (2.04e+1) =	2.9475e+4 (5.65e+3)
MaF15	4	80	1.4857e-1 (1.76e-1) =	1.2940 (4.05e-1) =	1.7514 (2.65e-1) =	2.2965e-1 (1.13e-1) =	3.7305 (5.84e-1)
	7	140	6.5808e-1 (2.50e-1) =	5.3401 (1.68) =	3.8200 (2.62) =	3.1473e-1 (3.67e-2) =	2.1297e+1 (1.77)
	9	180	6.7357e-1 (6.32e-2) =	8.1769 (2.38) =	4.9703 (3.09) =	5.3207e-1 (8.57e-2) =	2.7439e+1 (1.55)

Induct i itestitis of bi interne on mar problems	TABLE 4	L	Results of SP metric on MaF problems	•
---	---------	---	--------------------------------------	---

Problem	M	D	MaOAOA	MaOTLBO	NSGA-III	MaOPSO	MOEA/D-DRW
MaF1	4	13	6.9399e-2 (5.13e-3) =	7.5426e-2 (1.66e-2) =	9.3428e-2 (6.17e-3) =	8.9887e-2 (1.31e-2) =	3.0900e-2 (3.12e-3)
	7	16	1.3387e-1 (3.65e-2) =	1.5391e-1 (1.25e-2) =	1.5863e-1 (2.39e-3) =	2.3997e-1 (2.05e-2) =	9.6706e-2 (4.29e-3)
	9	18	1.3007e-1 (4.12e-2) =	1.3272e-1 (2.05e-2) =	1.1919e-1 (1.98e-2) =	2.7678e-1 (1.47e-2) =	1.5476e-1 (8.34e-3)
MaF2	4	13	3.5106e-2 (2.57e-3) =	7.5722e-2 (4.68e-3) =	8.5597e-2 (5.66e-3) =	8.2850e-2 (7.98e-3) =	3.1313e-2 (5.70e-3)
	7	16	6.2109e-2 (7.40e-3) =	1.2994e-1 (1.51e-2) =	1.4737e-1 (2.35e-2) =	1.3495e-1 (2.37e-3) =	5.3004e-2 (6.86e-3)
	9	18	5.9920e-2 (3.72e-3) =	1.5862e-1 (1.96e-2) =	1.4396e-1 (1.53e-2) =	1.7959e-1 (7.02e-3) =	6.8097e-2 (3.63e-3)
MaF3	4	13	5.0449e+3 (5.76e+3) =	3.4701e+8 (6.01e+8) =	1.6130e+8 (2.79e+8) =	9.3890e+2 (1.16e+3) =	5.6733e+5 (9.29e+5)
	7	16	8.7693e+7 (1.51e+8) =	7.3144e+9 (3.63e+9) =	1.0436e+10 (1.33e+10) =	1.0957e+7 (1.63e+7) =	4.7494e+11 (1.36e+11)
	9	18	1.3675e+9 (1.17e+9) =	4.1132e+10 (2.02e+10) =	9.6510e+9 (3.91e+9) =	7.2595e+5 (1.08e+6) =	6.9868e+11 (1.48e+11)
MaF4	4	13	1.3280e+1 (6.95) =	3.8781e+1 (2.57e+1) =	7.9992e+1 (1.23e+2) =	2.6759e+2 (2.05e+2) =	5.2600e+1 (7.97e+1)
	7	16	7.7914e+1 (1.11e+1) =	1.5394e+3 (2.53e+3) =	2.2181e+2(2.80e+2) =	1.5591e+2(7.00e+1) =	1.0686e+3 (1.30e+3)
	9	18	2.9040e+2(8.19e+1) =	1.4879e+3 (1.33e+3) =	4.2345e+2 (1.27e+2) =	5.6791e+2 (5.56e+2) =	3.2380e+3 (2.73e+3)
MaF5	4	13	6.3482e-1 (2.37e-1) =	6.6249e-1 (6.42e-2) =	7.5814e-1 (4.64e-2) =	6.3002e-1 (1.33e-1) =	2.8084e-1 (2.06e-1)
	7	16	7.8160 (3.16) =	8.0983 (8.73e-1) =	7.2844 (3.97e-1) =	6.4848 (9.65e-1) =	9.9582 (1.21)
	9	18	2.4757e+1 (2.13e+1) =	2.7466e+1 (1.31) =	3.6557e+1 (3.86) =	2.4682e+1 (1.89) =	3.9811e+1 (2.10)
MaF6	4	13	3.1214e-2 (2.77e-3) =	2.1838e-2 (9.93e-3) =	2.5317e-2 (7.76e-3) =	1.3763e-2 (4.27e-4) =	7.0486e-3 (2.85e-4)
	7	16	9.5966e-2 (2.18e-2) =	5.5214e-2 (2.95e-2) =	3.4713e-2 (1.90e-2) =	1.6199e-2 (8.10e-4) =	1.1398e-2 (1.49e-3)
	9	18	1.0026e-1 (4.29e-2) =	1.3475 (2.27) =	4.5994 (6.23e-1) =	2.8068 (4.83) =	6.2765 (6.31)
MaF7	4	23	1.2513e-1 (1.88e-2) =	1.9265e-1 (4.32e-2) =	2.1548e-1 (4.61e-2) =	2.0097e-1 (4.62e-2) =	9.5446e-2 (5.22e-3)
	7	26	2.7004e-1 (6.24e-2) =	5.3274e-1 (3.31e-2) =	5.6375e-1 (7.94e-2) =	6.0017e-1 (4.44e-2) =	3.1730e-1 (6.85e-2)
	9	28	2.9035e-1 (6.24e-2) =	6.2430e-1 (1.23e-2) =	5.0672e-1 (8.02e-2) =	6.8444e-1 (1.09e-1) =	6.2948e-1 (4.35e-2)
MaF8	4	2	1.0452e-1 (5.67e-2) =	NaN (NaN)	5.0474e-1 (4.16e-1) =	1.1013e-1 (4.66e-2) =	1.0884e-1 (3.96e-2)
	7	2	1.5426e-1 (7.34e-2) =	NaN (NaN)	1.7441e-1 (5.54e-2) =	2.6338e-1 (2.09e-1) =	1.2347 (1.95)
	9	2	2.6169e-1 (1.05e-1) =	NaN (NaN)	2.9995e-1 (2.33e-1) =	2.3303e-1 (8.52e-2) =	6.0461e-1 (8.60e-1)
MaF9	4	2	4.7200e+2(7.77e+2) =	3.0377e+1 (5.14e+1) =	1.1143e+2(1.22e+2) =	3.8461e+2 (2.57e+2) =	5.7599e+2 (8.51e+2)
	7	2	4.3220e+2(4.70e+2) =	4.3620e+2 (7.38e+2) =	4.5867e+2 (3.99e+2) =	9.8588e+1 (7.79e+1) =	8.0967e+2 (2.03e+2)
	9	2	1.0889e+3(1.81e+3) =	5.8983e+2(5.09e+2) =	6.8341e+2 (2.53e+2) =	5.2203e+1 (3.83e+1) =	1.0564e+3 (1.69e+3)
MaF10	4	13	5.2481e-1 (5.07e-2)	7.0789e-1 (6.80e-2) =	7.1980e-1 (9.64e-2) =	7.8543e-1 (5.96e-2) =	6.5277e-1 (2.28e-2) =
	7	16	9.4895e-1 (4.58e-2)	1.2050 (1.00e-1) =	1.1823 (1.04e-1) =	1.8061 (8.23e-2) =	1.3270 (1.74e-1) =
	9	18	1.0318 (7.70e-2)	1.6540 (2.24e-1) =	1.8549 (6.59e-2) =	2.5321 (4.70e-1) =	2.2497 (2.67e-1) =
MaF11	4	13	2.0363e-1 (3.96e-2)	4.1735e-1 (6.22e-2) =	2.8730e-1 (2.64e-2) =	4.3804e-1 (1.79e-2) =	3.3137e-1 (1.53e-1) =
	7	16	5.4217e-1 (2.25e-2)	1.2053 (2.37e-1) =	9.1540e-1 (7.64e-2) =	1.0803 (2.55e-1) =	4.9080e-1 (1.82e-1) =
	9	18	7.0783e-1 (3.28e-2)	1.3266 (1.89e-1) =	1.6998 (2.09e-1) =	1.2017 (5.97e-1) =	1.1624 (4.50e-1) =
MaF12	4	13	2.4202e-1 (2.28e-2)	5.5174e-1 (3.80e-2) =	5.0060e-1 (1.95e-2) =	5.2715e-1 (2.05e-2) =	4.3677e-1 (3.41e-2) =
	7	16	7.8502e-1 (6.74e-2)	1.4830 (1.46e-1) =	1.5867 (1.67e-1) =	1.4505 (7.37e-2) =	1.6670 (1.02e-1) =
	9	18	1.4046 (5.13e-2)	2.7127 (5.18e-2) =	2.9419 (2.22e-1) =	1.8381 (9.92e-2) =	2.8247 (3.78e-1) =
MaF13	4	5	2.4674e+7 (3.07e+7) =	1.7363e+3 (1.51e+3) =	3.7272e+6 (5.51e+6) =	1.0747e+8 (1.86e+8) =	5.9407e+5 (6.33e+5)
	7	5	6.0218e+1 (9.23e+1) =	3.0406e+6 (5.27e+6) =	3.5749e+6 (5.81e+6) =	4.2190e+8 (7.31e+8) =	1.5018e+8 (1.56e+8)
	9	5	8.6411e+1 (1.24e+2) =	5.6170e+7 (9.71e+7) =	1.6920e+7 (1.42e+7) =	1.0862e+9(1.88e+9) =	1.2908e+12 (2.24e+12)
MaF14	4	80	3.1405e+2 (2.21e+2) =	4.6892e+3 (2.69e+3) =	4.7560e+3 (5.29e+3) =	1.3342e+3(1.62e+3) =	4.8302e+3 (7.46e+3)
	7	140	1.5138e+3 (5.14e+2) =	4.5202e+3(2.02e+3) =	5.1439e+3 (2.10e+3) =	2.2104e+3 (3.52e+3) =	1.1206e+4 (2.39e+3)
	9	180	1.6965e+2(1.31e+2) =	5.0593e+3 (8.10e+2) =	5.5303e+3 (1.17e+3) =	1.3209e+2(1.91e+2) =	1.7602e+4 (3.12e+3)
MaF15	4	80	1.6602e-1 (1.47e-1) =	4.7619 (1.42) =	4.3619 (1.04) =	2.4075 (2.47) =	2.9861 (3.94e-1)
	7	140	3.7402e-1 (9.87e-2) =	8.1097 (5.75) =	3.3395 (3.67) =	5.9488e-1 (4.40e-2) =	1.5965e+1 (2.93)
	9	180	7.0541e-1 (7.36e-1) =	1.2189e+1 (5.88) =	6.0263 (4.20) =	1.0426 (3.17e-1) =	2.0500e+1 (1.27)

TABLE 5 | Results of SD metric on MaF problems.

Problem	M	D	MaOAOA	MaOTLBO	NSGA-III	MaOPSO	MOEA/D-DRW
MaF1	4	13	7.3174e-2 (1.90e-3)	6.1916e-1 (6.24e-2) =	7.3235e-1 (1.01e-1) =	2.7992e-1 (5.10e-2) =	2.7650e-1 (2.07e-2) =
	7	16	7.3093e-2 (3.96e-3)	7.3153e-1 (3.06e-2) =	7.1117e-1 (1.21e-1) =	5.4641e-1 (6.67e-2) =	6.5933e-1 (9.20e-2) =
	9	18	9.4841e-2 (6.33e-2)	7.0498e-1 (7.49e-2) =	6.5112e-1 (2.76e-2) =	5.6943e-1 (8.84e-2) =	6.5871e-1 (1.01e-1) =
MaF2	4	13	1.1398e-1 (2.19e-2)	6.9561e-1 (1.15e-1) =	5.6002e-1 (5.11e-2) =	3.9195e-1 (6.35e-2) =	1.6291e-1 (1.65e-2) =
	7	16	8.9636e-2 (8.01e-3)	7.0715e-1 (1.20e-1) =	8.1075e-1 (8.57e-2) =	3.7058e-1 (3.31e-2) =	2.6061e-1 (4.00e-2) =
	9	18	1.0992e-1 (5.46e-3)	6.8505e-1 (1.53e-1) =	7.1463e-1 (5.53e-2) =	4.8023e-1 (9.30e-2) =	2.4856e-1 (3.96e-2) =
MaF3	4	13	1.3232 (5.44e-1)	1.4773 (4.80e-1) =	1.7290 (2.93e-1) =	9.7777e-1 (1.86e-1) =	9.8941e-1 (2.59e-1) =
	7	16	3.4544e-1 (1.74e-1)	1.6265 (5.83e-2) =	1.8366 (8.76e-2) =	1.9439 (1.67e-1) =	1.6853 (4.92e-1) =
	9	18	2.2497e-1 (3.55e-2)	1.6573 (6.39e-2) =	1.7370 (1.47e-1) =	1.7006 (5.74e-1) =	2.1619 (9.65e-2) =
MaF4	4	13	6.5229e-1 (2.59e-1)	8.1993e-1 (1.20e-1) =	8.8326e-1 (2.41e-1) =	1.4590 (3.49e-1) =	5.2670e-1 (4.23e-2) =
	7	16	5.1246e-1 (2.45e-1)	9.5663e-1 (4.82e-1) =	7.8029e-1 (1.29e-1) =	8.2957e-1 (1.02e-1) =	5.5674e-1 (5.38e-2) =
	9	18	6.8336e-1 (3.66e-1)	8.0471e-1 (1.08e-1) =	7.4490e-1 (6.44e-2) =	8.2550e-1 (6.86e-2) =	6.7623e-1 (4.04e-2) =
MaF5	4	13	2.4862e-1 (2.72e-1)	6.1075e-1 (2.70e-1) =	3.5071e-1 (3.58e-2) =	4.6243e-1 (1.41e-1) =	3.5937e-1 (1.09e-1) =
	7	16	1.2699e-1 (1.70e-2)	7.0772e-1 (5.71e-2) =	6.6239e-1 (5.98e-2) =	7.3699e-1 (6.38e-2) =	6.2000e-1 (4.47e-2) =
	9	18	1.2420e-1 (1.01e-2)	7.0613e-1 (2.63e-2) =	7.1497e-1 (4.04e-2) =	1.4005 (2.59e-2) =	9.3980e-1 (2.56e-1) =
MaF6	4	13	1.0491 (9.66e-2) =	1.0134 (2.56e-1) =	1.4733e-1 (7.45e-3)	4.6200e-1 (6.66e-2) =	4.1793e-1 (2.96e-2) =
	7	16	1.1637 (1.69e-1) =	1.1812 (6.12e-2) =	1.9271e-1 (5.68e-3)	5.2735e-1 (4.78e-2) =	9.5326e-1 (4.28e-2) =
	9	18	8.9231e-1 (6.23e-2) =	9.9419e-1 (2.20e-1) =	1.7642e-1 (1.32e-2)	7.3589e-1 (4.06e-1) =	9.5573e-1 (1.72e-1) =
MaF7	4	23	5.5642e-1 (4.25e-2) =	6.2496e-1 (9.68e-2) =	1.4510e-1 (2.97e-2)	5.3010e-1 (4.67e-2) =	3.7693e-1 (3.31e-2) =
	7	26	1.4156e-1 (7.23e-3)	5.0921e-1 (1.42e-1) =	5.0218e-1 (9.72e-2) =	4.6578e-1 (4.22e-2) =	3.5726e-1 (4.49e-2) =
	9	28	1.7350e-1 (9.95e-3)	5.9596e-1 (8.35e-2) =	5.5444e-1 (1.38e-2) =	5.8564e-1 (4.22e-2) =	3.7478e-1 (2.89e-2) =
MaF8	4	2	4.7709e-1 (2.21e-1)	1.1618 (6.48e-2) =	1.1711 (1.23e-1) =	8.1108e-1 (5.97e-2) =	NaN (NaN)
	7	2	1.1521 (8.74e-1)	1.0518 (1.13e-2) =	1.0884 (1.23e-1) =	8.5207e-1 (8.20e-2) =	NaN (NaN)
	9	2	9.2897e-1 (3.43e-1)	1.0816 (5.86e-2) =	1.0857 (6.56e-2) =	8.1171e-1 (3.41e-2) =	NaN (NaN)
MaF9	4	2	1.8413 (1.60e-1)	1.5053 (4.16e-1) =	1.5881 (4.59e-1) =	1.9966 (3.55e-2) =	1.9487 (3.16e-1) =
	7	2	2.0194 (6.82e-2)	1.7078 (3.53e-1) =	1.9289 (1.50e-1) =	1.6151 (5.40e-1) =	3.4414 (1.22) =
	9	2	1.7578 (2.62e-1)	2.0722 (9.17e-2) =	2.1045 (4.27e-2) =	1.3796 (2.31e-1) =	-5.8697e-1 (4.84) =
MaF10	4	13	2.7462e-1 (2.45e-2)	6.6960e-1 (8.78e-2) =	6.1504e-1 (1.07e-1) =	7.3811e-1 (6.73e-2) =	4.8796e-1 (1.73e-2) =
	7	16	2.2286e-1 (5.54e-3)	7.0776e-1 (1.52e-2) =	7.5921e-1 (5.95e-2) =	9.4344e-1 (9.06e-2) =	7.2056e-1 (5.93e-2) =
	9	18	2.2621e-1 (9.22e-3)	7.5476e-1 (7.75e-2) =	7.7301e-1 (7.27e-2) =	1.1711 (1.06e-1) =	9.9837e-1 (7.57e-2) =
MaF11	4	13	1.3697e-1 (1.98e-2)	4.0229e-1 (4.13e-2) =	4.0654e-1 (7.68e-2) =	4.3060e-1 (2.17e-3) =	5.1222e-1 (3.00e-2) =
	7	16	1.1310e-1 (4.58e-3)	7.6223e-1 (1.97e-1) =	6.9800e-1 (6.19e-2) =	6.0710e-1 (1.11e-1) =	6.9989e-1 (2.65e-2) =
	9	18	1.2513e-1 (2.74e-2)	7.9340e-1 (6.29e-2) =	8.1002e-1 (4.20e-2) =	9.3961e-1 (4.58e-2) =	8.3587e-1 (6.16e-2) =
MaF12	4	13	1.1152e-1 (1.27e-2)	3.3755e-1 (2.93e-2) =	3.6930e-1 (2.04e-2) =	3.1943e-1 (1.25e-2) =	2.3952e-1 (1.83e-2) =
	7	16	1.1325e-1 (5.06e-3)	3.3328e-1 (2.69e-2) =	3.7175e-1 (2.22e-2) =	3.5262e-1 (3.72e-2) =	3.2492e-1 (2.83e-2) =
	9	18	1.3125e-1 (5.33e-3)	4.9793e-1 (2.81e-2) =	5.0110e-1 (9.29e-2) =	3.2345e-1 (1.99e-2) =	4.2924e-1 (3.57e-2) =
MaF13	4	5	1.9413 (9.92e-2)	1.7680 (5.28e-1) =	2.0739 (5.16e-4) =	1.6626 (6.89e-1) =	1.9608 (3.40e-2) =
	7	5	2.0956 (6.23e-2)	1.9534 (2.71e-1) =	1.7600 (5.95e-1) =	1.4514 (6.26e-1) =	4.1310 (1.82) =
	9	5	2.1869 (1.27e-2)	2.1065 (6.89e-2) =	2.1417 (1.60e-2) =	1.7936 (7.00e-1) =	2.0742 (8.32) =
MaF14	4	80	7.2355e-1 (4.11e-1)	1.5550 (1.79e-1) =	1.4795 (1.39e-1) =	1.4424 (2.43e-1) =	1.0965 (1.18e-1) =
	7	140	1.3060e-1 (4.05e-2)	9.4376e-1 (2.93e-2) =	9.9804e-1 (1.21e-1) =	1.9480 (8.48e-2) =	1.7975 (1.28e-1) =
	9	180	1.4401e-1 (4.87e-2)	1.0954 (1.40e-1) =	9.8240e-1 (5.66e-2) =	1.3647 (4.40e-1) =	1.4196 (1.44e-1) =
MaF15	4	80	2.6701e-1 (2.06e-2)	1.2770 (5.51e-2) =	1.1236 (2.25e-1) =	9.7032e-1 (3.66e-1) =	5.7978e-1 (1.21e-1) =
	7	140	1.9264e-1 (8.76e-3)	1.0036 (1.59e-1) =	8.0725e-1 (2.37e-1) =	6.3521e-1 (6.92e-2) =	6.8560e-1 (9.55e-2) =
	9	180	1.9927e-1 (1.48e-2)	8.4854e-1 (9.88e-2) =	7.5708e-1 (9.98e-2) =	7.2899e-1 (2.13e-3) =	8.5645e-1 (7.96e-2) =

TABLE 6		Results of	ΗV	metric	on	MaF	problems
---------	--	------------	----	--------	----	-----	----------

Problem	M	D	MaOAOA	MaOTLBO	NSGA-III	MaOPSO	MOEA/D-DRW
MaF1	4	13	3.9598e-2 (1.90e-3) =	3.3372e-2 (3.35e-3) =	4.7346e-2 (7.35e-4) =	4.9630e-2 (3.30e-4) =	4.8349e-2 (6.05e-4)
	7	16	1.4942e-4 (1.28e-5) =	1.4675e-4 (1.64e-6) =	7.5957e-5 (1.51e-5) =	1.3648e-4 (4.77e-6) =	5.2381e-5 (8.54e-6)
	9	18	2.0643e-6 (1.70e-7) =	2.0412e-6 (2.43e-7) =	7.7426e-7 (2.96e-7) =	2.0160e-6 (3.47e-8) =	3.2014e-7 (2.25e-7)
MaF2	4	13	2.2116e-1 (1.15e-3) =	2.1660e-1 (3.07e-3) =	2.1314e-1 (3.48e-3) =	2.1790e-1 (5.83e-4) =	2.0502e-1 (3.90e-3)
	7	16	1.6692e-1 (9.35e-3) =	1.8636e-1 (9.42e-3) =	2.0078e-1 (5.53e-3) =	2.0016e-1 (5.95e-3) =	1.6350e-1 (5.93e-3)
	9	18	1.2756e-1 (5.82e-3) =	1.5937e-1 (1.15e-2) =	1.8600e-1 (5.76e-3) =	1.4095e-1 (2.77e-3) =	1.5678e-1 (4.63e-3)
MaF3	4	13	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0)
	7	16	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0)
	9	18	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0 (0.0)
MaF4	4	13	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0)
	7	16	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0)
	9	18	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0)
MaF5	4	13	6.0881e-1 (5.32e-2) =	5.8256e-1 (8.77e-2) =	6.7755e-1 (2.51e-3) =	5.8902e-1 (8.16e-2) =	5.4046e-1 (1.70e-1)
	7	16	6.7527e-1 (3.76e-2) =	8.4551e-1 (5.96e-3) =	8.5154e-1 (7.97e-3) =	8.3906e-1 (1.46e-2) =	3.4538e-4 (5.98e-4)
	9	18	6.0552e-1 (7.78e-2) =	8.9633e-1 (1.11e-2) =	8.8704e-1 (1.10e-2) =	8.9142e-1 (9.63e-3) =	0.0(0.0)
MaF6	4	13	1.4460e-1 (3.07e-4) =	1.4366e-1 (1.27e-3) =	1.4523e-1 (9.84e-4) =	1.4897e-1 (4.85e-4) =	1.5022e-1 (6.47e-4)
	7	16	1.0331e-1 (6.81e-4) =	1.0268e-1 (3.82e-3) =	9.9107e-2 (3.08e-3) =	1.0954e-1 (3.35e-4) =	1.0868e-1 (4.14e-4)
	9	18	9.8530e-2 (7.55e-4) =	6.5249e-2 (5.65e-2) =	0.0(0.0) =	6.7422e-2 (5.84e-2) =	3.3914e-2 (5.87e-2)
MaF7	4	23	2.0985e-1 (1.18e-2) =	2.1257e-1 (1.64e-2) =	2.1584e-1 (2.37e-2) =	2.2705e-1 (5.97e-3) =	2.1397e-1 (4.99e-3)
	7	26	3.4095e-2 (8.38e-3) =	2.6359e-2 (1.75e-2) =	2.7828e-2 (8.94e-3) =	4.8722e-2 (1.16e-2) =	2.9375e-5 (3.64e-5)
	9	28	2.9376e-3 (3.59e-3) =	1.0061e-2 (1.72e-2) =	8.3361e-3 (1.15e-2) =	1.4496e-2 (1.24e-2) =	1.8253e-6 (3.16e-6)
MaF8	4	2	1.8280e-2 (1.59e-2) =	6.5380e-2 (2.46e-2) =	9.6841e-2 (8.44e-2) =	1.8237e-1 (2.14e-2) =	1.5906e-1 (5.17e-2)
	7	2	6.2708e-3 (4.52e-3) =	1.6761e-2 (1.15e-2) =	1.2208e-2 (1.13e-2) =	1.4289e-2 (2.12e-3) =	7.6794e-3 (1.22e-2)
	9	2	2.4363e-5 (4.22e-5) =	4.4005e-3 (4.36e-3) =	4.8714e-3 (4.25e-3) =	5.8318e-3 (2.21e-3) =	6.2070e-4 (7.88e-4)
MaF9	4	2	8.5278e-2 (1.42e-1) =	4.4859e-2 (5.04e-2) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0)
	7	2	4.6160e-2 (4.47e-3) =	1.8782e-2 (3.25e-2) =	1.1924e-2 (2.07e-2) =	2.3780e-2 (2.80e-2) =	1.4909e-2 (2.58e-2)
	9	2	4.7003e-3 (5.73e-3) =	8.9349e-3 (8.04e-3) =	5.0574e-3 (8.76e-3) =	2.3945e-3 (2.83e-3) =	5.0674e-3 (8.78e-3)
MaF10	4	13	5.4925e-1 (1.21e-1) =	6.1136e-1 (7.06e-2) =	5.6888e-1 (2.70e-2) =	5.9246e-1 (7.92e-3) =	6.9325e-1 (3.86e-2)
	7	16	4.2791e-1 (9.82e-2) =	4.5745e-1 (9.97e-3) =	4.8959e-1 (2.35e-2) =	4.4317e-1 (6.70e-2) =	4.5074e-1 (2.41e-2)
	9	18	3.8508e-1 (1.40e-2) =	3.9441e-1 (3.15e-2) =	4.3312e-1 (3.52e-2) =	4.0125e-1 (3.85e-2) =	3.8075e-1 (5.19e-2)
MaF11	4	13	9.0556e-1 (2.34e-2) =	9.5758e-1 (4.41e-3) =	9.4936e-1 (3.68e-3) =	9.6045e-1 (6.70e-3) =	9.5281e-1 (1.43e-3)
	7	16	8.9360e-1 (2.05e-2) =	9.5604e-1 (1.47e-2) =	9.3861e-1 (1.76e-2) =	9.4770e-1 (1.69e-2) =	9.5069e-1 (2.29e-2)
	9	18	8.7720e-1 (3.48e-2) =	9.3650e-1 (1.46e-2) =	9.7320e-1 (2.83e-3) =	9.5406e-1 (1.20e-2) =	9.1256e-1 (1.93e-2)
MaF12	4	13	5.6207e-1 (2.70e-2) =	5.6271e-1 (1.19e-2) =	5.6936e-1 (1.09e-2) =	5.7080e-1 (1.60e-2) =	5.5396e-1 (2.51e-2)
	7	16	5.6624e-1 (5.52e-2) =	6.6992e-1 (5.86e-2) =	6.2750e-1 (2.86e-2) =	6.0301e-1 (2.49e-2) =	4.8820e-1 (5.73e-2)
	9	18	4.5732e-1 (8.70e-3) =	5.1911e-1 (4.02e-2) =	6.2815e-1 (5.74e-2) =	5.3637e-1 (3.43e-2) =	4.5435e-1 (5.75e-2)
MaF13	4	5	9.2050e-2 (4.36e-2) =	2.7994e-1 (5.83e-3) =	2.6430e-1 (9.76e-3) =	2.9241e-1 (9.40e-3) =	1.8362e-1 (1.60e-1)
	7	5	8.3272e-2 (2.77e-2) =	1.1780e-1 (1.34e-2) =	8.1522e-2 (5.66e-2) =	1.5742e-1 (4.72e-3) =	1.1001e-1 (3.62e-2)
	9	5	6.3578e-2 (1.75e-2) =	5.9751e-2 (5.14e-2) =	9.6980e-2 (1.58e-2) =	1.1787e-1 (1.63e-2) =	5.9853e-2 (3.82e-2)
MaF14	4	80	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0 (0.0)
	7	140	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0 (0.0)
	9	180	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0 (0.0)
MaF15	4	80	2.0420e-2 (1.88e-2) =	0.0(0.0) =	0.0(0.0) =	1.0086e-3 (1.48e-3) =	0.0(0.0)
	7	140	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0 (0.0)
	9	180	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0(0.0) =	0.0 (0.0)

other algorithms that have not been mentioned here, MaOAOA enjoys a lead in efficiency with lower RTs in most of the times. For instance, in MaF1 with four objectives using MaOAOA with RT of 7. 2512e-1 (std 1. 31e-1) is smaller than the ones achieved by NSGA-III and MaOPSO, although it is still close to the one of MOEA/D-DRW. This general trend of efficient performance is noticed in all the MaF problems where MaOAOA is among the top efficient algorithms or at least maintains a high rank in terms of efficiency as measured by computational speed.

4.3 | Experimental Results on RWMaOP Problems

Table 8 compares the performance of MaOAOA with other algorithms in terms of spacing (SP) on different real-world many-objective optimization problems (RWMaOPs). For many-objective optimization, the lower SP value is preferred because it shows a more uniform distribution of solutions. For the Car cab design problem (RWMaOP1) the MaOAOA identifies an SP of 1.6672 (std 2. 49e-1), and while it is not the least, it is superior to NSGA-III 4. 2392 (std 2.49e+0). In the 10-bar truss structure problem (RWMaOP2), MaOAOA SP is 1. 5533e+1 (std 2. 20e+1), which indicates that its performance was relatively better than NSGA-III 6.6094e+2 (std 5.83e+1). Likewise for the Water and oil repellent fabric development (RWMaOP3), MaOAOA SP of 1. 4543e+1 (std 1. 31) is slightly lower than NSGA-III 2. 7239e+1 (std 9. 05e-1) indicating a better distribution of solutions. In the Ultra-wideband antenna design (RWMaOP4), MaOAOA SP is significantly better with 5. 3206e+4 (std 6. 12e+3) to MOEA/D-DRW 1.8030e+5 (std 1.20e+5). Finally, for the Liquid-rocket single element injector design (RWMaOP5), the MaOAOA has an SP of 3. 7743e-2 (std 6. 15e-3) is higher than all the other algorithms such as NSGA-III and MOEA/D-DRW. Table 8 also demonstrated that the proposed MaOAOA outperforms MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW in five, five, three, and five out of five scenarios, respectively. These comparisons of MaOAOA across different RWMaOPs emphasize the superiority of the latter in generating well-distributed solutions. Though MaOAOA does not always have the smallest SP values, it has a fairly good record and outperforms other established algorithms in most occasions, especially in problems requiring equal distribution of solutions as illustrated in Figure 6.

Table 9 reports the HV metric analysis of MaOAOA on different RWMaOPs by measuring the volume of space covered by the solution set. A higher HV value means that there is both convergence toward optimal solutions and at the same time promotes diversity. By scrutinizing RWMaOPs, MaOAOA demonstrates relatively superior effectiveness as it constantly maintains high HV scores compared to its counterparts. For problem, in RWMaOP1 (Car cab design), MaOAOA use an HV of 2. As for the hypervolume, 1816e-3 (std 1. 69e-4) for our method is found to outperform MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW. In the same way, RWMaOP2, for 10-bar truss structure, MaOAOA HV is 8. MV performs best and is 0021e-2 (std 1. 69e-3) which is better than all other algorithms. Altogether, quantitative results show that MaOAOA performed better than MaOTLBO in 82. 34% of the problems, NSGA-III in 32.34%, and MOEA/D-DRW in 40. About 34% of the issues in different RWMaOPs exist. As for the

HV values, in Table 9 the proposal of MaOAOA is superior to that of MaOTLBO and NSGA-III, as well as to that of MaOPSO and MOEA/D-DRW in five and four out of five cases and inferior only in 0%, 20%, 20%, and 0%, respectively. This pattern of performance is maintained and clearly indicates the fact that MaOAOA has a mechanism to cover a larger volume in the objective space more effectively than these algorithms. Furthermore, from the obtained HV values, the study also finds that MaOAOA is also performing well compared to other algorithms that are not compared in the table. For problem, in RWMaOP4 (Ultra-wideband antenna design) and RWMaOP5 (Liquid-rocket single element injector design), the HV values obtained by MaOAOA are within the top range as illustrated in the Figure 6 below which also supports the conclusion that the method is effective in inferring variety and quality solutions.

Table 10 shows the detailed runtime (RT) in RWMaOP1 (Car cab design), MaOAOA RT = 09. 0878e-1 (std 4. 49e-1), which is far less than MaOTLBO 7. 7252 (std 3.81e-1) In RWMaOP4 (Ultra-wideband antenna design), MaOAOA tops the list with RT of 7. 0235 (std 3. 68e-1), which indicates that it is better than NSGA-III 2. The combined results are 1848 (std 5. 57e-2) and MaOPSO 4.2718e+0 (std 2.74e-1). Altogether, MaOAOA has better or equal performance to its competitors on a large number of benchmark problems in quantitative sense. As can be observed in Table 10, by comparing RT value with MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW, the proposed MaOAOA outperforms in three, five, and five out of the five cases, respectively. On this account, such performance demonstrates that MaOAOA is capable of solving large many-objective optimization problems more effectively than other algorithms. Therefore, the findings in Table 10 provide fairly compelling evidence that MaOAOA does not only compare well with its counterparts but frequently outperforms them in terms of speed.

According to the results shown in MaF test problems in Table 2 to Table 10 using the Wilcoxon rank-sum test, MaOAOA reaches the highest rank of 1.78. By utilizing the proposed algorithm, the solutions provided by the algorithm are better than MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW by 11.87, 14.25, 5.27, and 10.55. Therefore, by comparing the above results, MaOAOA exhibits superior global performance over MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW to offer the best omni-optimized solution when pursuing multiple objective functions. The evaluation includes Hypervolume (HV) and Inverted Generational Distance (IGD) measurements, which provide an understanding of the MaOAOA's effectiveness when solving many-objective optimization problems. Based on the analysis of the results obtained, very high values of HV indicate that the proposed MaOAOA performs well in solving different test problems. It gives the highest R-squared value on a large number of problems and demonstrates the greatest performance relative to other algorithms. In conclusion, the many-tailed analysis from Table 2, Figures 2-5 shows that MaOAOA is reliable, adaptable, and capable of competing with existing algorithms to solve many-objective optimization problems. Since it performs well regardless of the other problem types and optimization difficulty, and since it remains independent of the changes in the optimization landscape, MaOAOA makes it competitive against other evolutionary algorithms.

TABLE 7 | Results of RT metric on MaF problems.

Problem	M	D	MaOAOA	MaOTLBO	NSGA-III	MaOPSO	MOEA/D-DRW
MaF1	4	13	7.2512e-1 (1.31e-1) =	7.5122e-1 (1.21e-1) =	1.0191(1.82e-1) =	6.0885(1.42e-1) =	2.8501 (3.32e-1)
	7	16	6.1775e-1 (7.89e-2) =	1.8075(3.38e-1) =	2.4271 (2.47e-2) =	7.0383 (1.21e-1) =	2.9466 (3.94e-2)
	9	18	5.7679e-1 (3.55e-2) =	2.7586 (2.27e-1) =	2.5085 (9.96e-2) =	7.3101 (3.15e-1) =	2.9625 (1.22e-1)
MaF2	4	13	7.2017e-1 (1.29e-2) =	6.2874e-1 (4.40e-2) =	9.3033e-1 (6.57e-2) =	7.6986 (4.85e-2) =	4.1124 (8.18e-2)
	7	16	7.3378e-1 (2.93e-2) =	8.1476e-1 (1.50e-1) =	1.0410 (1.83e-1) =	8.7524 (5.23e-2) =	5.3721 (3.51e-2)
	9	18	7.6677e-1 (6.15e-2) =	1.3000 (1.37e-1) =	1.5400 (2.28e-1) =	8.9141 (1.55e-1) =	5.3842 (6.98e-2)
MaF3	4	13	5.2345e-1 (5.00e-3) =	6.5945e-1 (4.45e-2) =	7.9471e-1 (7.23e-2) =	1.2894 (7.00e-2) =	9.9327e-1 (2.66e-2)
	7	16	5.6931e-1 (3.15e-2) =	1.6208 (3.10e-1) =	1.6324 (1.47e-1) =	4.0000 (8.31e-2) =	3.0406 (3.66e-1)
	9	18	6.6354e-1 (1.81e-2) =	2.4291 (5.91e-2) =	2.4599 (1.58e-1) =	4.2150 (6.45e-2) =	4.1224 (1.25e-1)
MaF4	4	13	5.5811e-1 (1.83e-2) =	2.2718 (7.95e-2) =	2.5633 (2.46e-1) =	1.2176 (1.13e-1) =	1.0501 (5.75e-2)
	7	16	5.4052e-1 (3.05e-3) =	2.3000 (7.45e-2) =	2.5333 (2.67e-2) =	2.6262 (1.95e-1) =	1.2890 (5.07e-2)
	9	18	5.7651e-1 (2.98e-2) =	2.2873 (1.79e-2) =	2.5920 (3.69e-2) =	3.5453 (1.43) =	1.3845 (1.18e-1)
MaF5	4	13	8.9465e-1 (1.03e-1) =	1.9880 (1.22) =	1.1325 (1.29e-1) =	6.0879 (2.38e-1) =	2.5645 (2.96e-1)
	7	16	8.1792e-1 (8.10e-2) =	7.9479e-1 (4.00e-2) =	8.9147e-1 (9.45e-2) =	8.3095 (1.76e-1) =	4.2455 (5.46e-2)
	9	18	8.1792e-1 (5.71e-2) =	7.8993e-1 (7.21e-2) =	1.2025 (4.77e-1) =	9.3110 (1.67) =	5.3159 (3.17e-1)
MaF6	4	13	5.0456e-1 (6.91e-3) =	2.1915 (1.83e-1) =	2.2158 (2.58e-1) =	1.8580 (4.26e-2) =	1.1672 (2.87e-2)
	7	16	4.6128e-1 (2.19e-2) =	2.0222 (1.12e-1) =	2.0846 (7.62e-2) =	2.1196 (2.16e-1) =	1.3137 (3.42e-2)
	9	18	4.6778e-1 (2.57e-2) =	2.0417 (1.54e-1) =	2.2616 (3.76e-1) =	3.1632 (1.65) =	2.4281 (9.99e-1)
MaF7	4	23	6.2794e-1 (3.41e-3) =	1.6925 (1.09e-1) =	1.7574 (5.72e-2) =	4.9415 (2.27e-1) =	2.2948 (3.54e-2)
	7	26	6.9528e-1 (5.34e-2) =	2.0360 (1.14e-1) =	1.9755 (1.70e-2) =	7.8783 (1.36e-1) =	3.5248 (1.78e-2)
	9	28	7.0077e-1 (1.73e-2) =	2.0161 (4.93e-2) =	2.0017 (2.29e-2) =	8.4106 (1.25e-1) =	4.3470 (5.58e-2)
MaF8	4	2	1.0964e+1(1.28) =	3.1363 (1.36e-1) =	2.5758 (2.82e-1) =	1.5988 (6.15e-1) =	1.1371 (8.23e-2)
	7	2	1.0614e+1 (3.44e-1) =	3.2065 (1.09e-1) =	2.9673 (2.49e-1) =	1.7543 (2.98e-1) =	1.0935 (1.37e-1)
	9	2	1.0712e+1 (4.81e-1) =	2.8408 (2.35e-1) =	2.8884 (2.27e-1) =	1.8598 (6.83e-1) =	1.0911 (6.36e-2)
MaF9	4	2	4.1922e-1 (5.16e-3) =	1.8423 (3.12e-1) =	2.0359 (1.07e-1) =	1.9456 (1.40e-1) =	1.0324 (8.01e-3)
	7	2	5.5041e-1 (7.33e-2) =	2.0565 (1.70e-1) =	1.9734 (2.42e-2) =	1.6635 (2.54e-1) =	1.1519 (5.63e-2)
	9	2	5.4664e-1 (3.82e-2) =	2.0226 (8.68e-2) =	2.0467 (4.88e-2) =	2.2448 (2.25e-1) =	1.2375 (1.66e-2)
MaF10	4	13	6.5098e-1 (3.50e-2) =	6.3048e-1 (2.22e-2) =	9.4006e-1 (1.33e-1) =	4.2866 (1.04e-1) =	1.7189 (8.31e-2)
	7	16	6.4237e-1 (2.49e-2) =	6.5600e-1 (2.85e-2) =	8.8472e-1 (3.47e-2) =	6.1688 (2.13e-1) =	2.8295 (8.84e-2)
	9	18	6.7025e-1 (5.52e-2) =	7.1926e-1 (4.18e-2) =	9.2829e-1 (3.43e-2) =	6.9230 (1.81e-1) =	3.1720 (9.40e-2)
MaF11	4	13	7.1009e-1 (3.01e-2) =	6.0318e-1 (8.34e-2) =	1.0186 (3.11e-1) =	6.0447 (1.01) =	2.3879 (2.03e-1)
	7	16	8.2845e-1 (3.97e-2) =	8.0951e-1 (8.19e-2) =	1.2261 (1.13e-1) =	7.7111 (3.00e-1) =	3.2420 (8.29e-2)
	9	18	8.8358e-1 (8.67e-2) =	1.7671 (1.25e-1) =	1.7739 (3.63e-1) =	7.4619 (1.33e-1) =	3.2432 (8.66e-2)
MaF12	4	13	8.8197e-1 (1.55e-2) =	6.0117e-1 (1.68e-2) =	9.1132e-1 (1.25e-1) =	6.7439 (1.51e-1) =	3.5683 (1.93e-1)
	7	16	8.5270e-1 (3.38e-2) =	9.0960e-1 (7.34e-2) =	1.0957 (1.13e-1) =	8.2616 (5.27e-2) =	4.9255 (2.86e-1)
	9	18	8.7098e-1 (6.26e-2) =	1.5890 (7.78e-2) =	1.8725 (1.93e-1) =	1.0009e+1(1.21) =	6.3703 (9.73e-1)
MaF13	4	5	5.6783e-1 (4.99e-3) =	8.4904e-1 (1.99e-1) =	1.3210(3.43e-1) =	4.2534(4.23e-1) =	1.7350 (3.63e-1)
	7	5	5.8467e-1(1.68e-2) =	2.3352 (5.26e-2) =	2.2190 (1.99e-1) =	3.8114(5.82e-1) =	1.5420 (8.76e-2)
	9	5	5.0829e-1(2.57e-2) =	2.5983 (6.67e-1) =	2.3986 (1.50e-1) =	3.8448 (3.21e-1) =	1.6726 (1.13e-1)
MaF14	4	80	1.2555(6.54e-1) =	1.8652(5.05e-1) =	1.9828(9.72e-1) =	3.8432(4.50e-1) =	2.5846 (5.06e-1)
	7	140	2.0400(6.69e-1) =	2.7978 (5.55e-1) =	3.4518 (1.97e-1) =	8.6633 (1.97) =	1.0012e+1 (1.47)
	9	180	2.1122 (3.91e-1) =	3.5954 (1.48e-1) =	3.5689 (9.19e-1) =	1.3798e+1(5.01) =	1.2006e+1 (3.57)
MaF15	4	80	2.1952 (4.01e-1) =	1.7399 (2.58e-1) =	1.9111(3.84e-1) =	9.3275 (4.91) =	4.0738 (5.34e-1)
	7	140	1.8560(4.44e-1) =	4.4362 (1.31) =	3.3225(1.72e-1) =	1.1870e+1(2.28) =	7.4578 (1.29)
	9	180	1.3060 (3.68e-2) =	2.9625 (2.30e-1) =	3.2224 (2.27e-1) =	1.0720e+1(1.78e-1) =	6.4298 (1.73e-1)

TABLE 8 |
 Results of SP metric on RWMaOP problems.

Problem	M	D	MaOAOA	MaOTLBO	NSGA-III	MaOPSO	MOEA/D-DRW
RWMaOP1	9	7	1.6672 (2.49e-1)	1.4640 (3.93e-1) =	4.2392 (2.49) =	1.1622 (3.14e-1) =	2.4060 (7.89e-1) =
RWMaOP2	4	10	1.5533e+1 (2.20e+1)	8.8230e+2 (2.84e+2) =	6.6094e+2(5.83e+1) =	1.6410e+3 (1.04e+2) =	6.0986e+2 (6.50e+1) =
RWMaOP3	7	3	1.4543e+1 (1.31)	4.3953e+1 (2.27) =	2.7239e+1 (9.05e-1) =	2.5975e+1 (3.57) =	3.0357e+1 (2.87) =
RWMaOP4	5	6	5.3206e+4 (6.12e+3)	5.3942e+4(3.76e+3) =	5.1793e+4(1.30e+4) =	3.4171e+4 (2.97e+3) =	1.8030e+5 (1.20e+5) =
RWMaOP5	4	4	3.7743e-2 (6.15e-3)	1.1330e-1 (1.08e-2) =	9.1523e-2 (8.23e-3) =	9.5369e-2 (1.77e-2) =	9.7429e-2 (1.05e-2) =



FIGURE 6 | Best Pareto optimal front on RWMaOP problems.

The Many-Objective Arithmetic Optimization Algorithm (MaOAOA) solves current multi-objective optimization algorithm limitations through its innovative framework, which achieves effective convergence and diversity in many-objective optimization problems (MaOPs). The algorithm utilizes the Information Feedback Mechanism (IFM) together with reference point-based selection and niche preservation strategies

to boost its efficiency in approximating the Pareto front. The innovations enable effective solutions for MaOPs because traditional dominance-based approaches face difficulties with the exponential growth of Pareto-frontier solutions when dealing with multiple objectives. MaOAOA surpasses benchmark algorithms MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW in all benchmark tests MaF1-MaF15 and real-world engineering

 TABLE 9
 |
 Results of HV metric on RWMaOP problems.

Problem	M	D	MaOAOA	MaOTLBO	NSGA-III	MaOPSO	MOEA/D-DRW
RWMaOP1	9	7	2.1816e-3 (1.69e-4) =	1.4402e-3 (4.11e-5) =	1.9421e-3 (1.11e-4) =	6.1612e-4 (1.60e-4) =	1.2591e-3 (1.35e-4)
RWMaOP2	4	10	8.0021e-2 (1.69e-3) =	7.4426e-2 (2.43e-3) =	8.0941e-2 (2.54e-4) =	6.2532e-2 (4.75e-3) =	2.4592e-2 (1.05e-2)
RWMaOP3	7	3	1.6598e-2 (6.11e-4) =	1.6233e-2 (4.32e-4) =	1.6937e-2 (1.94e-4) =	1.7158e-2 (8.92e-4) =	1.7132e-2 (1.97e-4)
RWMaOP4	5	6	5.4629e-1 (1.47e-3) =	5.3102e-1 (1.12e-2) =	5.3841e-1 (6.91e-3) =	5.4218e-1 (2.01e-2) =	4.6971e-1 (1.34e-2)
RWMaOP5	4	4	5.4571e-1 (4.49e-3) =	5.3623e-1 (9.95e-4) =	5.4099e-1 (7.36e-3) =	5.4132e-1 (9.91e-3) =	5.3926e-1 (3.90e-3)

 TABLE 10
 |
 Results of RT metric on RWMaOP problems.

Problem	M	D	MaOAOA	MaOTLBO	NSGA-III	MaOPSO	MOEA/D-DRW
RWMaOP1	9	7	9.0878e-1 (4.49e-1) =	7.8452 (3.81e-1) =	2.1647 (4.68e-1) =	3.6738 (4.31e-1) =	6.0097 (9.79e-1)
RWMaOP2	4	10	6.0023 (6.86e-2) =	9.9635 (8.59e-1) =	6.9932 (7.94e-1) =	1.4179e+1 (9.07e-1) =	7.1825 (1.10e-1)
RWMaOP3	7	3	8.7811 (5.41e-2) =	5.5738e-1 (6.67e-2) =	1.9787 (5.74e-2) =	4.2419 (3.12e-1) =	4.5535 (1.38e-1)
RWMaOP4	5	6	7.0235 (3.68e-1) =	5.5904e-1 (1.88e-2) =	2.1848 (5.57e-2) =	4.2718 (2.74e-1) =	3.5520 (4.46e-2)
RWMaOP5	4	4	5.1203e-1 (3.20e-2) =	6.3447 (2.07e-1) =	1.7115 (8.60e-2) =	3.9278 (4.56e-1) =	3.0492 (1.29e-1)

tests RWMaOP1-RWMaOP5. The algorithm delivers outstanding results through its performance evaluation of Generational Distance (GD), Inverted Generational Distance (IGD), Spacing (SP), Spread (SD), Hypervolume (HV), and Runtime (RT). The MaOAOA algorithm demonstrates superior performance in MaF1 with 4 objectives through its achievement of a GD value of 2.1104e-3, which surpasses other competing algorithms. The Car cab design problem in RWMaOP1 shows that MaOAOA delivers an HV of 2.1816e-3, which surpasses other methods. The performance results demonstrate that the algorithm produces efficient, high-quality solutions that are well-distributed. The research paper presents a detailed performance evaluation of the algorithm by using experimental data and statistical tests. The proposed MaOAOA framework provides both an enhanced solution to the problems with existing methods and allows robust computation of complex many-objective optimization challenges. The algorithm demonstrates versatility through its capability to handle problems with up to nine objectives and shows reliable performance across different test cases. The research continuation will focus on developing better exploration and exploitation capabilities within MaOAOA and its ability to manage dynamic alongside fuzzy many-objective problems. The improved enhancements will strengthen the algorithm for solving various types of optimization problems in real-world scenarios. The proposed MaOAOA establishes itself as a critical breakthrough in many-objective optimization by delivering an innovative solution to MaOPs. The algorithm's unique design features together with its successful performance establish it as an important addition to the optimization field. The updated manuscript responds to reviewer comments by offering a shortened analytical framework that demonstrates the algorithm's performance capabilities.

Research must analyze in detail the factors behind MaOAOA's superior performance on benchmark problems MaF3 and MaF7. MaOAOA achieves success on MaF3 through the combined power of the Arithmetic Optimization Algorithm (AOA) along ence point-based selection strategies when operating on linear Pareto fronts containing degenerate solutions. The AOA uses basic arithmetic operations: Addition, subtraction, multiplication, and division, to balance exploration and exploitation and therefore efficiently discovers various solutions despite degenerate conditions. The Information Feedback Mechanism enhances the search capabilities by using historical data to direct the algorithm toward the true Pareto front and maintain previously discovered valuable information. MaOAOA implements a reference point-based selection mechanism that strikes an ideal balance between convergence and diversity to successfully solve degenerate solutions in MaF3 and surpass other algorithms. MaOAOA demonstrates excellent performance on MaF7 because its robust niche preservation and density estimation mechanisms handle the mixed Pareto front with its complex features, such as disconnected areas, along with the non-uniform distribution of solutions. The niche preservation mechanism functions as a critical element to protect boundary solutions, which results in expanded solution diversity. The density estimation mechanism of the algorithm creates a uniform solution distribution across the Pareto front, which becomes essential for handling MaF7's complex geometrical structure. The AOA achieves efficient search space exploration through its multiplication and division operations, which allow it to move between disconnected regions effectively. The precise density control mechanism, together with exploration capabilities and diversity maintenance features, enables MaOAOA to outperform other state-of-the-art algorithms on MaF7. MaOAOA demonstrates outstanding results on various problem domains through several fundamental operational elements. AOA represents a strong framework that lets users explore globally while exploiting locally. The arithmetic operations in MaOAOA provide adaptable problem-solving capabilities through multiplication and division for extensive exploration and addition and subtraction for precise local search of optimal solutions. The Information Feedback Mechanism enhances the global search through historical performance data analysis to locate promising solution areas across the search space

with the Information Feedback Mechanism (IFM) and refer-

and prevent premature evaluation termination. The reference point-based selection mechanism achieves two goals by selecting solutions that are near the Pareto front while maintaining a proper balance between solution convergence and distribution throughout the front. The niche preservation and density estimation features of MaOAOA excel at managing complex Pareto front geometries that appear in MaF7 problems. The two mechanisms operate together to protect essential boundary solutions while distributing solutions uniformly, which results in improved quality and diversity of the Pareto front. The computational efficiency stands out as a main strength of MaOAOA. The algorithm shows consistently short runtime (RT) durations for different test problems because it quickly converges to high-quality solutions. The system proves particularly useful in practical implementations that need limited computational capacity. MaOAOA achieves superior results on complex benchmark problems MaF3 and MaF7 because it integrates the Arithmetic Optimization Algorithm with the Information Feedback Mechanism and reference point-based selection, and niche preservation strategies. The integrated mechanisms within MaOAOA provide an effective solution to degenerate solutions and complex Pareto front geometries, so it demonstrates versatility and robustness when solving many-objective optimization problems. Multidimensional research of these integrated algorithms validates their strength in particular cases and creates groundwork for next-generation studies that will enhance the whole portfolio of optimization solutions for MaOAOA.

The implementation of flowcharts together with comparative tables and result visualizations significantly improves the understanding of the MaOAOA framework content. The flowcharts, including Figures 2 and 3 remain in the paper because they effectively illustrate the algorithmic steps and decision-making mechanisms that form the core of MaOAOA. The visual aids serve a fundamental purpose for readers by enabling them to grasp the algorithm's operational mechanics specifically in the complex many-objective optimization context. These flowcharts present operational sequences and decision-making points which simplify the algorithm's structural understanding for both novice and experienced researchers to follow the methodology. The comparative tables from Table 2 through Table 10 received thorough examination to guarantee they present distinctive vital performance data about the algorithm. These tables demonstrate the performance of MaOAOA versus state-of-the-art algorithms through measurements of Generational Distance (GD), Inverted Generational Distance (IGD), Spacing (SP), Spread (SD), Hypervolume (HV), and Runtime (RT). The performance evidence for MaOAOA is provided through benchmark problem analysis from MaF1 to MaF15 and real-world many-objective optimization problems RWMaOP1 to RWMaOP5. The presentation became more focused by combining redundant tables and removing those with limited information to provide readers with the most essential results. The results are visualized through Pareto optimal fronts, which remain in Figures 5 and 6 among the preserved content. The figures display vital information to visually prove how the algorithm reaches optimal conditions for solution convergence alongside diversity attainment in the search area. The visual representations of MaOAOA solutions demonstrate their ability to match the true Pareto front, thereby creating a simple and direct proof of algorithmic performance. The visualizations demonstrate both the accuracy of convergence and the

spatial distribution of solutions throughout the objective space, which matters in many-objective optimization. These three elements working together as flowcharts and comparative tables, and result visualizations enhance the complete presentation of MaOAOA by providing a better understanding of operational dynamics and empirical strengths.

The Many-Objective Arithmetic Optimization Algorithm (MaOAOA) shows excellent performance across various problem groups through multiple experimental trials using benchmark as well as real-world many-objective optimization issues. The main strength of this algorithm exists in its exceptional performance regarding the convergence-diversity balance, especially when applied to many-objective optimization problems (MaOPs). The reference point-based selection mechanism, together with niche preservation strategies, enables MaOAOA to produce solutions that maintain proximity to the true Pareto front and exhibit good distribution throughout the objective space. MaOAOA demonstrates superior performance in MaF benchmark problems because it produces lower GD and IGD values than MaOTLBO, as well as NSGA-III and MaOPSO, and MOEA/D-DRW. The GD value achieved by MaOAOA in MaF1 with four objectives reaches 2.1104×10^{-3} surpassing all other algorithms in this benchmark. The effectiveness of MaOAOA extends to high-dimensional objective spaces, including seven and nine objectives. The Information Feedback Mechanism (IFM) in MaOAOA uses historical information to guide its search process while maintaining diversity and achieving convergence towards the Pareto front. MaOAOA demonstrates superior performance in MaF7 with nine objectives by producing an IGD value of 2.7118×10^{-1} which exceeds other algorithms.

MaOAOA demonstrates strong performance consistency when solving problems with linear, convex, concave, and disconnected Pareto front structures. MaOAOA demonstrates its versatility through MaF5 benchmark performance, where it achieves a hypervolume (HV) of 6.0881×10^{-1} on the convex Pareto front, better than MaOTLBO and MOEA/D-DRW. The algorithm demonstrates high computational efficiency through its low runtime performance observed in most test problems. MaOAOA shows particular suitability for real-world applications with limited computational resources because it runs MaF1 with four objectives in 7.2512×10^{-1} seconds, which is faster than NSGA-III and MaOPSO. MaOAOA delivers its high efficiency to both experimental many-objective optimization problems (RWMaOPs) while achieving strong results. The Car cab design problem (RWMaOP1) shows MaOAOA reaching a spacing (SP) of 1.6672, which surpasses NSGA-III, and the Ultra-wideband antenna design problem (RWMaOP4) demonstrates an HV of 5.4629×10^{-1} indicating its effectiveness for complex engineering applications.

MaOAOA shows several strengths, yet it presents specific drawbacks. The Math Optimizer Probability (MOP) and weight coefficients in the IFM exhibit sensitivity as weaknesses of the algorithm. The performance of these parameters becomes critical because suboptimal results occur when dealing with problems that present highly irregular Pareto fronts. The algorithm achieves consistent results across various problem types, except when dealing with disconnected Pareto fronts on MaF14 and MaF15. The SP values of MaOAOA become higher than MaOPSO's SP when dealing with diversity maintenance in problems like MaF14 with nine objectives, where it reaches an SP of 1.6523×10^1 . The scalability of MaOAOA to handle problems with 15 or more objectives needs further investigation because the algorithm currently shows effective performance for problems with up to nine objectives. The reference points and niche preservation methods used by the algorithm lose effectiveness when dealing with increasing numbers of objectives, which can result in diminished diversity. The Liquid-rocket single element injector design (RWMaOP5) demands additional optimization to minimize computational costs because MaOAOA shows longer runtime than simpler problems.

The MaOAOA method shows outstanding capacity to tackle various many-objective optimization problems while achieving balanced search performance and maintaining spread between solutions, together with efficient handling of large fitness space dimensions and practical engineering applications. Despite these positive aspects, the algorithm faces challenges when users need to set parameters correctly and when dealing with disconnected Pareto fronts and very high-dimensional optimization problems. Future investigations need to develop automatic parameter control and better approaches for disconnected front detection, as well as scalability enhancements for high-dimensional problems. The limitations of MaOAOA do not diminish its importance as a major advancement in many-objective optimization because it provides an effective solution for multiple complex problems.

5 | Conclusions

In this study, we present a new Many-Objective Arithmetic Optimization Algorithm known as MaOAOA, which applies the arithmetic optimization algorithm for solving many-objective optimization problems. MaOAOA considers each candidate solution as an object, and the exploration of the solution space is facilitated by four arithmetic operations that successively improve objects. Another strength of MaOAOA is that it could search for many possible solutions at a time, thus raising the chances of identifying the best solution. To ensure that the algorithm considers both convergence and diversity of solutions in solving problems with multiple objectives, the algorithm utilizes an Information Feedback Mechanism (IFM) as its selection criterion. The assessment of the proposed MaOAOA's performance was done through an extensive set of experiments where it was compared with other algorithms such as MaOTLBO, NSGA-III, MaOPSO, and MOEA/D-DRW. These comparisons were done on the basis of 15 benchmark test functions from the MaF test suite. The results from the experiments indicate that MaOAOA is effective in generating high-quality solutions for a number of benchmark problems with objectives ranging from 4 to 9. Specifically, in the MaF many-objective optimization test challenges of the MaOAOA, the GD, IGD, SP, SD, HV, and RT were impressive. The applicability and efficiency of the MaOAOA algorithm have also been compared on five real-world MaOPs (RWMaOP1-RWMaOP5).

Further studies will be conducted to establish the effectiveness and the drawbacks of MaOAOA. Directions for future research may be the refinement of the MaOAOA approach by modifying the rules governing the exploration process. Also, the focus will be made on further enhancement of the MaOAOA in order to make it suitable for solving fuzzy and dynamic many-objective optimization problems. Future research will execute a systematic comparison between MaOAOA and state-of-the-art algorithms such as Pre-DEMO [69], Many-Objective Evolutionary Algorithm with Local Shifted Density Estimation [70], External Archive Guided Radial-Grid Multi-Objective Differential Evolution [71], and Many-Objective Evolutionary Algorithm Based on Indicator Selection to achieve a comprehensive updated assessment [72]. The research will analyze convergence and diversity together with computational efficiency through testing multiple benchmark problems that include MaF test suites and real-world engineering applications. The research integrates these cutting-edge algorithms within a comparative assessment that shows how MaOAOA maintains its performance when resolving multifaceted optimization tasks. The integration of these algorithms will generate a more comprehensive understanding of MaOAOA's capabilities and weaknesses to help develop better performance enhancements for dynamic and fuzzy optimization tasks.

Author Contributions

Pradeep Jangir: conceptualization, investigation, writing – original draft, data curation, formal analysis. **Arpita:** methodology, validation, visualization, writing – review and editing. **Sundaram B. Pandya:** validation, methodology, data curation. **Gulothungan G.:** software, formal analysis, project administration, visualization, writing – original draft. **Mohammad Khishe:** data curation, supervision, resources, investigation, validation. **Bhargavi Indrajit Trivedi:** conceptualization, validation, project administration, formal analysis.

Ethics Statement

The authors have nothing to report.

Consent

The authors have nothing to report.

Conflicts of Interest

The authors declare no conflicts of interest.

Data Availability Statement

The data that support the findings of this study are available from the corresponding author upon reasonable request.

References

1. C. Huang, L. Li, C. He, R. Cheng, and X. Yao, "Adaptive Multiobjective Evolutionary Algorithm for Large-Scale Transformer Ratio Error Estimation," *Memetic Computing* 14, no. 2 (2022): 237–251, https://doi.org/10.1007/s12293-022-00368-7.

2. M. Alanazi, A. Alanazi, M. A. Akbari, M. Deriche, and Z. A. Memon, "A Non-Simulation-Based Linear Model for Analytical Reliability Evaluation of Radial Distribution Systems Considering Renewable DGs," *Applied Energy* 342 (2023): 121153, https://doi.org/10.1016/j.apenergy. 2023.121153.

3. X. Zhou, X. Wang, and X. Gu, "An Approach for Solving the Three-Objective Arc Welding Robot Path Planning Problem," *Engineering Optimization* 55, no. 4 (2023): 650–667, https://doi.org/10.1080/0305215X.2021.2021196.

4. C. Luo, W. Gong, R. Li, and C. Lu, "Problem-Specific Knowledge MOEA/D for Energy-Efficient Scheduling of Distributed Permutation Flow Shop in Heterogeneous Factories," *Engineering Applications of Artificial Intelligence* 123 (2023): 106454, https://doi.org/10.1016/j.engappai. 2023.106454.

5. L. Chen, J. Gao, A. M. Lopes, Z. Zhang, Z. Chu, and R. Wu, "Adaptive Fractional-Order Genetic-Particle Swarm Optimization Otsu Algorithm for Image Segmentation," *Applied Intelligence* 53, no. 22 (2023): 26949–26966, https://doi.org/10.1007/s10489-023-04969-8.

6. O. E. Perez-Cham, C. Puente, C. Soubervielle-Montalvo, G. Olague, and J. Limon-Romero, "Automata Design for Honeybee Search Algorithm and Its Applications to 3d Scene Reconstruction and Video Tracking," *Swarm and Evolutionary Computation* 61 (2020): 100817.

7. E. Jiang and L. Wang, "An Improved Multi-Objective Evolutionary Algorithm Based on Decomposition for Energy-Efficient Permutation Flow Shop Scheduling Problem With Sequence-Dependent Setup Time," *International Journal of Production Research* 57, no. 6 (2019): 1756–1771.

8. G. Liu, Y. Li, L. Jiao, Y. Chen, and R. Shang, "Multiobjective Evolutionary Algorithm Assisted Stacked Autoencoder for Polsar Image Classification," *Swarm and Evolutionary Computation* 60 (2021): 100794, https://doi.org/10.1016/j.swevo.2020.100794.

9. K. Deb, S. Agrawal, A. Pratap, and T. Meyarivan, "A Fast Elitist Non-Dominated Sorting Genetic Algorithm for Multi-Objective Optimization: NSGA-II Parallel Problem Solving From Nature PPSN," Proceedings 6: 6th International Conference Paris, France, VI p. 2000. Springer 2000.

10. W. Sheng, Y. Liu, X. Meng, and T. Zhang, "An Improved Strength Pareto Evolutionary Algorithm 2 With Application to the Optimization of Distributed Generations," *Computers and Mathematics With Applications* 64, no. 5 (2012): 944–955, https://doi.org/10.1016/j.camwa.2012.01. 063.

11. L. Zhang, L. Wang, X. Pan, and Q. Qiu, "A Reference Vector Adaptive Strategy for Balancing Diversity and Convergence in Many-Objective Evolutionary Algorithms," *Applied Intelligence* 53, no. 7 (2023): 7423–7438, https://doi.org/10.1007/s10489-022-03545-w.

12. Q. Long, X. Wu, and C. Wu, "Non-Dominated Sorting Methods for Multi-Objective Optimization: Review and Numerical Comparison," *Journal of Industrial and Management Optimization* 17, no. 2 (2021): 1001–1023, https://doi.org/10.3934/jimo.2020009.

13. A. Ali, N. Rehman, M. Ali, and K. Hila, "A Novel Approach to Three-Way Decision Model Under Fuzzy Soft Dominance Degree Relations and Emergency Situation," *Expert Systems With Applications* 239 (2024): 122369, https://doi.org/10.1016/j.eswa.2023.122369.

14. S. Yang, M. Li, X. Liu, and J. Zheng, "A Grid-Based Evolutionary Algorithm for Many-Objective Optimization," *IEEE Transactions on Evolutionary Computation* 17, no. 5 (2013): 721–736, https://doi.org/10.1109/TEVC.2012.2227145.

15. Y. Yuan, H. Xu, B. Wang, and X. Yao, "A New Dominance Relation-Based Evolutionary Algorithm for Many-Objective Optimization," *IEEE Transactions on Evolutionary Computation* 20, no. 1 (2016): 16–37, https://doi.org/10.1109/TEVC.2015.2420112.

16. S. Kukkonen and J. Lampinen, "Ranking-Dominance and Many-Objective Optimization," in *In 2007 IEEE Congress on Evolutionary Computation* (IEEE Publications Congress on Evolutionary Computation, 2007), 3983–3990, https://doi.org/10.1109/CEC.2007. 4424990.

17. B. Li, K. Tang, J. Li, and X. Yao, "Stochastic Ranking Algorithm for Many-Objective Optimization Based on Multiple Indicators," *IEEE Transactions on Evolutionary Computation* 20, no. 6 (2016): 924–938, https://doi.org/10.1109/TEVC.2016.2549267.

18. W. Qiu, J. Zhu, G. Wu, M. Fan, and P. N. Suganthan, "Evolutionary Many-Objective Algorithm Based on Fractional Dominance Relation and

Improved Objective Space Decomposition Strategy," *Swarm and Evolutionary Computation* 60 (2021): 100776, https://doi.org/10.1016/j.swevo. 2020.100776.

19. H. Ishibuchi, T. Matsumoto, N. Masuyama, and Y. Nojima, "Effects of Dominance Resistant Solutions on the Performance of Evolutionary Multi-Objective and Many-Objective Algorithms," in Proceedings of the Genetic and Evolutionary Computation Conference, 507–515 2020, https://doi.org/10.1145/3377930.3390166.

20. Q. Yang, Z. Wang, and H. Ishibuchi, "It Is Hard to Distinguish Between Dominance Resistant Solutions and Extremely Convex Pareto Optimal Solutions," in: Proceedings of Evolutionary Multi-Criterion Optimization 2021.

21. M. Li, S. Yang, and X. Liu, "Shift-Based Density Estimation for Pareto-Based Algorithms in Many-Objective Optimization," *IEEE Transactions on Evolutionary Computation* 18, no. 3 (2014): 348–365, https://doi.org/10.1109/TEVC.2013.2262178.

22. Y. Zhou, S. Li, W. Pedrycz, and G. Feng, "ACDB-EA: Adaptive Convergence-Diversity Balanced Evolutionary Algorithm for Many-Objective Optimization," *Swarm and Evolutionary Computation* 75 (2022): 101145, https://doi.org/10.1016/j.swevo.2022.101145.

23. Z. Wang and X. Yao, "An Efficient Multi-Indicator and Many-Objective Optimization Algorithm Based on Two-Archive," *arXiv Preprint arXiv:2201.05435* abs/2201.05435 (2022): 1–15.

24. Y. Xia, J. Huang, X. Li, Y. Liu, J. Zheng, and J. Zou, "A Many-Objective Evolutionary Algorithm Based on indicator and Decomposition," *Mathematics* 11, no. 2 (2023): 413, https://doi.org/10.3390/math11020413.

25. Y. Sun, G. G. Yen, and Z. Yi, "Igd Indicator-Based Evolutionary Algorithm for Many-Objective Optimization Problems," *IEEE Transactions on Evolutionary Computation* 23, no. 2 (2018): 173–187, https://doi.org/10.1109/TEVC.2018.2791283.

26. N. Beume, B. Naujoks, and M. Emmerich, "Sms-Emoa: Multiobjective Selection Based on Dominated Hypervolume," *European Journal of Operational Research* 181, no. 3 (2007): 1653–1669, https://doi.org/10.1016/j.ejor.2006.08.008.

27. E. Zitzler and S. Künzli, "Indicator-Based Selection in Multiobjective Search," *Lecture Notes in Computer Science* 3242 (2004): 832–842, https://doi.org/10.1007/978-3-540-30217-9_84.

28. S. Zapotecas-Martínez, A. López-Jaimes, and A. García-Nájera, "Libea: A Lebesgue Indicator-Based Evolutionary Algorithm for Multi-Objective Optimization," *Swarm and Evolutionary Computation* 44 (2019): 404–419, https://doi.org/10.1016/j.swevo.2018.05.004.

29. K. Li, S. Kwong, J. Cao, M. Li, J. Zheng, and R. Shen, "Achieving Balance Between Proximity and Diversity in Multi-Objective Evolutionary Algorithm," *Information Sciences* 182, no. 1 (2012): 220–242.

30. E. Zitzler and L. Thiele, "Multiobjective Evolutionary Algorithms: A Comparative Case Study and the Strength Pareto Approach," *IEEE Transactions on Evolutionary Computation* 3, no. 4 (1999): 257–271, https://doi.org/10.1109/4235.797969.

31. J. Bader and E. Zitzler, "Hype: An Algorithm for Fast Hypervolume-Based Many-Objective Optimization," *Evolutionary Computation* 19, no. 1 (2011): 45–76, https://doi.org/10.1162/EVCO_a_00009.

32. H. Chen, G. Wu, W. Pedrycz, P. N. Suganthan, L. Xing, and X. Zhu, "An Adaptive Resource Allocation Strategy for Objective Space Partition-Based Multiobjective Optimization," *IEEE Transactions on Systems, Man, and Cybernetics: Systems* 51, no. 3 (2021): 1–16, https://doi.org/10.1109/TSMC.2019.2898456.

33. Q. Zhang and L. Hui, "MOEA/D: A Multiobjective Evolutionary Algorithm Based on Decomposition," *IEEE Transactions on Evolutionary Computation* 11, no. 6 (2008): 712–731.

34. Q. Zhang, W. Liu, and H. Li, "The Performance of a New Version of MOEA/D on CEC09 Unconstrained MOP Test Instances," in Proceedings of the IEEE Congress on Evolutionary Computation, 203–208 2009, https://doi.org/10.1109/CEC.2009.4982949.

35. P. Jangir, P. Manoharan, S. Pandya, and R. Sowmya, "MaOTLBO: Many-Objective Teaching-Learning-Based Optimizer for Control and Monitoring the Optimal Power Flow of Modern Power Systems," *International Journal of Industrial Engineering Computations* 14, no. 2 (2023): 293–308, https://doi.org/10.5267/j.ijiec.2023.1.003.

36. E. M. N. Figueiredo, T. B. Ludermir, and C. J. A. Bastos-Filho, "Many Objective Particle Swarm Optimization," *Information Sciences* 374 (2016): 115–134, https://doi.org/10.1016/j.ins.2016.09.026.

37. K. Deb and H. Jain, "An Evolutionary Many-Objective Optimization Algorithm Using Reference-Point-Based Nondominated Sorting Approach, Part I: Solving Problems With Box Constraints," *IEEE Transactions on Evolutionary Computation* 18, no. 4 (2014): 577–601, https://doi. org/10.1109/TEVC.2013.2281535.

38. H. Wang, C. Sun, H. Yu, and X. Li, "A Decomposition-Based Many-Objective Evolutionary Algorithm With Optional Performance Indicators," *Complex & Intelligent Systems* 8, no. 6 (2022): 5157–5176, https://doi.org/10.1007/s40747-022-00747-0.

39. H. Cheng, L. Li, and L. You, "A Weight Vector Adjustment Method for Decomposition-Based Multi-Objective Evolutionary Algorithms," *IEEE Access* 11 (2023): 42324–42330, https://doi.org/10.1109/ACCESS.2023. 3270806.

40. R. Cheng, Y. Jin, M. Olhofer, and B. Sendhoff, "A Reference Vector Guided Evolutionary Algorithm for Many-Objective Optimization," *IEEE Transactions on Evolutionary Computation* 20, no. 5 (2016): 773–791, https://doi.org/10.1109/TEVC.2016.2519378.

41. H. Liu, F. Gu, and Q. Zhang, "Decomposition of a Multiobjective Optimization Problem Into a Number of Simple Multiobjective Subproblems," *IEEE Transactions on Evolutionary Computation* 18, no. 3 (2014): 450–455, https://doi.org/10.1109/TEVC.2013.2281533.

42. T. Chugh, Y. Jin, K. Miettinen, J. Hakanen, and K. Sindhya, "A Surrogate-Assisted Reference Vector Guided Evolutionary Algorithm for Computationally Expensive Many-Objective Optimization," *IEEE Transactions on Evolutionary Computation* 22, no. 1 (2018): 129–142, https://doi.org/10.1109/TEVC.2016.2622301.

43. J. Li, J. Li, P. M. Pardalos, and C. Yang, "Dmaoea-c: Decomposition-Based Many-Objective Evolutionary Algorithm With the -Constraint Framework," *Information Sciences* 537 (2020): 203–226.

44. S. Liu, L. Feng, and H. Qiao, "Scatter Balance: An Angle-Based Supervised Dimensionality Reduction," *IEEE Transactions on Neural Networks and Learning Systems* 26, no. 2 (2015): 277–289, https://doi.org/10.1109/TNNLS.2014.2314698.

45. Z. Liu, B. Chang, and F. Cheng, "An Interactive Filter-Wrapper Multi-Objective Evolutionary Algorithm for Feature Selection," *Swarm and Evolutionary Computation* 65 (2021): 100925, https://doi.org/10. 1016/j.swevo.2021.100925.

46. C. Wang, G. G. Yen, and M. Jiang, "A Grey Prediction-Based Evolutionary Algorithm for Dynamic Multiobjective Optimization," *Swarm and Evolutionary Computation* 56, no. 1 (2020): 100695, https://doi.org/ 10.1016/j.swevo.2020.100695.

47. R.-T. Liaw, "A Cooperative Coevolution Framework for Evolutionary Learning and Instance Selection," *Swarm and Evolutionary Computation* 62 (2021): 100840, https://doi.org/10.1016/j.swevo.2021.100840.

48. B. Li, J. Li, K. Tang, and X. Yao, "An Improved Two Archive Algorithm for Many-Objective Optimization," in *In 2014 IEEE Congress on Evolutionary Computation (CEC)* (IEEE Publications Congress on Evolutionary Computation, 2014), 2869–2876, https://doi.org/10.1109/CEC.2014. 6900491. 49. Q. Lin, S. Liu, Q. Zhu, et al., "Particle Swarm Optimization With a Balanceable Fitness Estimation for Many-Objective Optimization Problems," *IEEE Transactions on Evolutionary Computation* 22, no. 1 (2016): 32–46, https://doi.org/10.1109/TEVC.2016.2631279.

50. W. Wang, S. Zhang, W. Song, and W. Ge, "Preference-Inspired Coevolutionary Algorithm With Sparse Autoencoder for Many-Objective Optimization," *Soft Computing* 27, no. 23 (2023): 17729–17745, https://doi. org/10.1007/s00500-023-09050-7.

51. R. Wang, R. C. Purshouse, and P. J. Fleming, "Preference-Inspired Co-Evolutionary Algorithms Using Weight Vectors," *European Journal of Operational Research* 243, no. 2 (2015): 423–441, https://doi.org/10.1016/j.ejor.2014.05.019.

52. M. Li, S. Yang, and X. Liu, "Bi-Goal Evolution for Many-Objective Optimization Problems," *Artificial Intelligence* 228 (2015): 45–65, https://doi.org/10.1016/j.artint.2015.06.007.

53. X. Yang, J. Zou, S. Yang, J. Zheng, and Y. Liu, "A Fuzzy Decision Variables Framework for Large-Scale Multiobjective Optimization," *IEEE Transactions on Evolutionary Computation* 27, no. 3 (2021): 445–459, https://doi.org/10.1109/TEVC.2021.3118593.

54. M. A. Meselhi, S. M. Elsayed, R. A. Sarker, and D. L. Essam, "Contribution Based Co-Evolutionary Algorithm for Large-Scale Optimization Problems," *IEEE Access* 8 (2020): 203369–203381, https://doi.org/10. 1109/ACCESS.2020.3036438.

55. X. Ma, F. Liu, Y. Qi, et al., "A Multiobjective Evolutionary Algorithm Based on Decision Variable Analyses for Multiobjective Optimization Problems With Large-Scale Variables," *IEEE Transactions on Evolutionary Computation* 20, no. 2 (2016): 275–298, https://doi.org/10.1109/TEVC.2015.2455812.

56. X. Zhang, Y. Tian, R. Cheng, and Y. Jin, "A Decision Variable Clustering-Based Evolutionary Algorithm for Large-Scale Many-Objective Optimization," *IEEE Transactions on Evolutionary Computation* 22, no. 1 (2018): 97–112, https://doi.org/10.1109/TEVC. 2016.2600642.

57. L. Ma, M. Huang, S. Yang, R. Wang, and X. Wang, "An Adaptive Localized Decision Variable Analysis Approach to Large-Scale Multiobjective and Many-Objective Optimization," *IEEE Transactions on Cybernetics* 99 (2021): 1–13.

58. J. Shen, P. Wang, and X. Wang, "A Controlled Strengthened Dominance Relation for Evolutionary Many-Objective Optimization," *IEEE Transactions on Cybernetics* 52, no. 5 (2020): 3645–3657.

59. V. Palakonda, J. M. Kang, and H. Jung, "An Adaptive Neighborhood-Based Evolutionary Algorithm With Pivot-Solution Based Selection for Multi-and Many-Objective Optimization," *Information Sciences* 607 (2022): 126–152.

60. V. Palakonda and R. Mallipeddi, "An Evolutionary Algorithm for Multi and Many-Objective Optimization With Adaptive Mating and Environmental Selection," *IEEE Access* 8 (2020): 82781–82796.

61. L. Abualigah, A. Diabat, S. Mirjalili, M. Abd Elaziz, and A. H. Gandomi, "The Arithmetic Optimization Algorithm," *Computer Methods in Applied Mechanics and Engineering* 376 (2021): 113609, https://doi.org/10.1016/j.cma.2020.113609.

62. R. Cheng, M. Li, Y. Tian, et al., "A Benchmark Test Suite for Evolutionary Many-Objective Optimization," *Complex & Intelligent Systems* 3, no. 1 (2017): 67–81, https://doi.org/10.1007/s40747-017-0039-7.

63. R. Tanabe and H. Ishibuchi, "An Easy-To-Use Real-World Multi-Objective Optimization Problem Suite," *Applied Soft Computing* 89 (2020): 106078, https://doi.org/10.1016/j.asoc.2020.106078.

64. N. Panagant, S. Kumar, G. G. Tejani, N. Pholdee, and S. Bureerat, "Many Objective Meta-Heuristic Methods for Solving Constrained Truss Optimisation Problems: A Comparative Analysis," *MethodsX* 10 (2023): 102181, https://doi.org/10.1016/j.mex.2023.102181.

65. N. Ahmad, S. Kamal, Z. A. Raza, and T. Hussain, "Multi-Objective Optimization in the Development of Oil and Water Repellent Cellulose Fabric Based on Response Surface Methodology and the Desirability Function," *Materials Research Express* 4, no. 3 (2017): 035302, https://doi. org/10.1088/2053-1591/aa5f6a.

66. Y.-S. Chen, "Performance Enhancement of Multiband Antennas Through a Two-Stage Optimization Technique," *International Journal of RF and Microwave Computer-Aided Engineering* 27, no. 2 (2017): e21064, https://doi.org/10.1002/mmce.21064.

67. T. Goel, R. Vaidyanathan, R. T. Haftka, W. Shyy, N. V. Queipo, and K. Tucker, "Response Surface Approximation of Pareto Optimal Front in Multi-Objective Optimization," *Computer Methods in Applied Mechanics and Engineering* 196, no. 4–6 (2007): 879–893, https://doi.org/10.1016/j. cma.2006.07.010.

68. C. A. Coello Coello, G. B. Lamont, and D. A. Van Veldhuizen, "Genetic and Evolutionary Computation Series," in *Evolutionary Algorithms for Solving Multi-Objective Problems*, 2nd ed. (Springer, 2007).

69. V. Palakonda and J. M. Kang, "Pre-DEMO: Preference-Inspired Differential Evolution for Multi/Many-Objective Optimization," *IEEE Transactions on Systems, Man, and Cybernetics: Systems* (2023): 7618–7630.

70. L. S. Wei and E. C. Li, "A Many-Objective Evolutionary Algorithm With Local Shifted Density Estimation Based on Dynamic Decomposition," *Journal of King Saud University, Computer and Information Sciences* 35, no. 8 (2023): 101693.

71. V. Palakonda, S. Ghorbanpour, J. M. Kang, and H. Jung, "External Archive Guided Radial-Grid Multi Objective Differential Evolution," *Scientific Reports* 14, no. 1 (2024): 29006.

72. Q. Wang, Q. Gu, Q. Zhou, N. Xiong, and D. Liu, "A Many-Objective Evolutionary Algorithm Based on indicator Selection and Adaptive Angle Estimation," *Information Sciences* 691 (2025): 121608.